



A Comprehensive Evaluation of the Structural Stability of Core-Shell Hybrid Nanostructures for Hybrid Supercapacitor Applications

S.E Umoru¹

¹Dept. of Natural Science, National Institute for Nigerian Languages, Aba, Nigeria

Corresponding Author: efemenaumoru@gmail.com

Received: 18/Feb/2024; Accepted: 20/Mar/2024; Published: 30/Apr/2024

Abstract—With a focus on hybrid supercapacitor applications, this thorough study attempts to present an in-depth evaluation of the status of research on the structural stability of core-shell hybrid nanostructures. In order to summarize important discoveries, identify trends, and draw attention to obstacles in the way of improving the structural stability of these nanostructures, the review looks at a broad variety of investigations that have been done in this field of research. Since core-shell hybrid nanostructures combine a functional shell with a stable core material, they present a promising material for enhancing the longevity and performance of hybrid supercapacitors. The review examines the advantages that these nanostructures provide, such as increased electrical conductivity, longer cycle life, and stronger mechanical strength. It also looks at the fundamental of hybrid supercapacitors, classes of core-shell hybrid nanostructures, types of core-shell hybrid nanostructures, factors that affect the structural stability of core-shell hybrid nanostructures, and the strategies for enhancing the structural stability of core-shell hybrid nanostructures under various operating conditions, such as mechanical stress, humidity, and temperature fluctuations, requires more research. For these nanostructures to be widely used in real-world supercapacitor applications, it is also necessary to address their scalability and cost-effectiveness.

Keywords—Comprehensive evaluation, Structural stability, Core-shell hybrid nanostructures, Durability, cycling stability, Hybrid supercapacitor, Energy storage, Performance.

1. Introduction

The modern world is becoming more and more dependent on efficient and sustainable power sources, which makes the development of energy storage devices crucial [1]. One of the most promising alternatives for energy storage systems is the hybrid supercapacitor, which combines the advantages of batteries with ordinary supercapacitors. The design of the hybrid supercapacitor is crucial since the type of electrode material used determines how effectively the device functions. The development of novel materials and device designs that enhance energy storage and performance has been the main objective of hybrid supercapacitor research. Researchers have looked into a range of materials in an attempt to boost hybrid supercapacitors' capacitance. Metal oxides, conductive polymers, carbon-based compounds, and mesoporous electrode materials are a few of these materials. Graphene, carbon nanotubes, and activated carbon are examples of carbon-based materials with high surface area and electrical conductivity that are widely used as supercapacitor electrode materials [2]. However, their limitations concerning energy density and specific

capacitance limit their overall performance in hybrid supercapacitors [3]. Conversely, conductive polymers, like polythiophene, polyaniline, polypyrrole, and polyacetylene, are well known for having a large capacitance in comparison to their high conductivity [4]. However, the reduced cycle stability and mechanical instability of conductive polymers have adversely affected their short-term performance as electrode materials in hybrid supercapacitors [5]. Transition metal oxides, such as ruthenium oxide, manganese dioxide, nickel oxide, and vanadium pentoxide, are used as pseudocapacitive materials in supercapacitor electrodes [6]. Despite having a high specific capacitance, transition metal oxides' poor electrical conductivity and limited cycle stability have an impact on their overall performance in hybrid supercapacitors [7]. The potential of mesoporous materials, including mixed metal oxides, in energy storage applications is being investigated. However, issues with scalability and cost-effectiveness arise from their may synthesis. manufacturing processes, and use in supercapacitor electrodes [8]. Given the shortcomings of these electrode materials for hybrid supercapacitors, scientists are working harder to find alternative materials that may improve hybrid supercapacitors'

functionality. Core-shell hybrid nanostructures are one electrode material that has drawn interest because of its potential to enhance the functionality of hybrid supercapacitors.

Core-shell hybrid nanostructures have great potential, but structural stability is a major problem for them [9]. The issue is that existing core-shell hybrid nanostructures frequently experience structural degradation as a result of recurrent charge/discharge cycles, notwithstanding their efficiency in energy storage and charge/discharge cycles [10]. The hybrid supercapacitors' overall performance and lifespan may be shortened as a result of this structural instability. The restricted implementation of these core-shell hybrid nanostructures in real-world supercapacitor applications stems from the absence of an ideal method for augmenting their structural integrity. This threat must be addressed in order to fully exploit the potential of core-shell hybrid nanostructures and advance the creation of robust and more effective supercapacitors. Thus, a thorough analysis that concentrates on the difficulties related to the structural stability of coreshell hybrid nanostructures in hybrid supercapacitors is required. The objective of this review work is to pinpoint the root causes of structural deterioration and investigate viable approaches to improve stability. By finding a solution to this issue, scientists can progress energy storage technology by designing and creating hybrid supercapcitors that are more resilient and long-lasting. The foundations of hybrid supercapacitors will be covered in detail first, followed by an examination of core-shell technology, structural stability issues, methods for improving structural stability in core-shell hybrid nanostructures, and core-shell hybrid nanostructure processes, all of which will lead to a conclusion.

The rest part of the manuscript is organized as follows, Section 1 contains the introduction of of the study, Section 2 contain the review of related work of the study, section 3 contains some theories used by researchers in studying the structural stability of core-shell hybrid nanostructures, section 4 contains the method used in this review paper. Section 5 brings out the challenges associated with core-shell hybrid nanostructures, section 6 contains the findings on the methods for enhancing the structural stability of core-shell hybrid nanostructures, section 7 contains the conclusion and future scope.

2. Related work

2.1 Fundamental of Hybrid Supercapacitor

One kind of energy storage device that combines the benefits of batteries and supercapacitors is the hybrid supercapacitor. Because of its design, which provides superior power density, energy density (up to 10 times), and an operating voltage of 3.8 V compared to symmetric supercapacitors, it can be used in a wide range of applications. Hybrid supercapacitors are often composed of two primary electrodes: one that functions as a large energy storage battery and the other that functions as a powerful supercapacitor [11]. Hybrid supercapacitors are designed to perform better in terms of cycle life, efficiency, charge/discharge rates, and total energy storage capacity [12]. Double-layer capacitance and pseudocapacitance are the two main ways that hybrid supercapacitors work [13]. When electrical charges are trapped in an electric double layer between the electrode and the electrolyte, double-layer capacitance happens at the interface [14]. This method yields a constrained energy density but a high power density. Conversely, pseudocapacitance enables greater energy storage capacity and is predicated on reversible faradaic redox processes at the electrode-electrolyte interface [15]. The energy and power densities of several electrochemical energy storage systems are displayed in Figure 1.

Hybrid supercapacitors can balance high power and high energy storage by combining these two methods [17]. Maher et al. (2015) [18] demonstrated how to combine active materials with high-voltage electrolytes and logically construct the electrode microstructure to create threedimensional, high-performance hybrid supercapacitors and microsupercapacitors using graphene and MnO_2

Consequently, hybrid electrodes are generated that have a volumetric capacitance more than $1,100 \text{ F/cm}^3$.



conversion systems.[16].

Accordingly, the specific capacitance of the component MnO_2 is 1,145 F/g, very close to the expected value of 1,380 F/g. The overall energy density of the device varies from 22 to 42 Wh/l depending on its setup. This is superior to the energy densities of lithium-ion capacitors, hybrid supercapacitors, double-layer supercapacitors, and commercially available supercapacitors tested under the same conditions. It is comparable to lead acid batteries.

There are three different kinds of hybrid supercapacitors: selfcharging supercapacitors, asymmetric supercapacitors, and hybrid batteries/supercapacitors. The first and second variants are a combination of the two techniques that use electrostatic adsorption and Faradaic redox processes to simultaneously obtain high specific energy and specific power. Asymmetric supercapacitors are primarily made up of two different types

of electrodes: electrodes that are similar to batteries and electrodes that are similar to supercapacitors. Battery/supercapacitor hybrids are devices that combine batteries and supercapacitors in series through a variety of techniques. These consist of metal batteries [20], lithium ion batteries [21, 22], lead-acid batteries [19], and lithium sulfur batteries [23]. Moreover, self-charging Supercapacitors allow for the simultaneous harvesting and storing of many energy sources. The three different supercapacitor-based hybrid device types and their hybridization process are depicted in Figure 2. Hybrid supercapacitors have several potential applications in wearable, implantable, and portable electronics in the future energy storage systems.

The design and material selection of hybrid supercapacitors have a significant impact on their performance. In order to enhance the energy storage capacity of hybrid supercapacitors, scientists have investigated various combinations of electrode materials, electrolytes, and device designs. Research has been done on nanoporous designs and composite materials, for example, to increase the surface area and improve ion diffusion within the electrodes. Furthermore, the choice of electrolyte can have a significant effect on the performance of hybrid supercapacitors [24]. Aqueous electrolytes and ionic liquids are often used due to their superior stability and conductivity [25]. The voltage window, power densities, energy densities, and specific capacitance of the three types of hybrid devices-which employed CNT/PANI electrodes as inserting objects-that were previously mentioned are shown in Table 1. The Li-ion hybrid supercapacitor outperforms the others overall, which could make it a key candidate for additional study and



Figure. 2 : The three different hybrid device types that arise from the hybridization process of supercapacitors. Hybrid supercapacitors have a lot to offer future wearable, implantable, and portable electronics in terms of energy storage systems.[26].

 Table 1: lists of the three types of hybrid devices' potential windows, energy densities (Es), power densities (Ps), specific capacitance (Cs), and hybrid

teeninque.										
Device	Method	Cs	E _s	P _s	Pote	Refere				
	01	(mA n	(w n	(KW		nce				
	Hybrid	g ')	Kg ')	Kg ⁻)	wind					
					OW (J.)					
					(V)					
Lead-acid	Lead-	10 (19%	19.3	20.9	1.6-	27				
hybrid	acid	improve	(21%	×10-5	2.6					
Supercap	battery	ment)	improve							
acitor.	and		ment)							
	supercap									
	acitor									
	(outside)									
	in series									
7	7	22.4	47.0	14.1	0	20				
Zn–	Zn-	33.4	47.9	14.1	0-	28				
MnO ₂	MnO ₂	(1.5	(3.5		1.8					
nyoria	batteries	times	times							
Supercap	and	improve	improve							
actor.	supercap	ment)	ment)							
	action (in a ida)									
	(inside)									
Liton	in series	61.1	100	777	2.0	20				
LI-IOII hash ui d	Adding a	01.1	100	21.1	2.0-	29				
nyoria	CNIT/DA	(18%)	(22%)		4.1					
Supercap	CN1/PA	improve	improve							
actor.		ment)	ment)							
	electrode									
	into the									
	battery									

To optimize their electrochemical performance, core-shell hybrid nanostructures are used as the electrode material in hybrid supercapacitors. For example, the core material serves as a framework for the deposition of the active material, while the shell material aids in charge transfer and structural integrity. This design increases the materials' energy-storing capability by optimizing the use of the active materials. Using a simple one-pot synthesis technique based on a water-in-oil microemulsion system, Eunho Lim et al. (2015) [30] create Nb₂O₅@carbon core-shell nanocrystals (Nb₂O₅@C NCs) as high-power anode materials with controllable crystalline phases (orthorhombic (T) and pseudohexagonal (TT)). According to the authors, the T-Nb₂O₅@C NCs show a reversible specific capacity of about 180 mA h g-1 at 0.05 A g-1 (1.1-3.0 V against Li/Li+) with high rate capability, in contrast to TT-Nb₂O₅@C and carbon shell-free Nb₂O₅ NCs. The main reasons for this are (i) T-Nb₂O₅'s structural merit and (ii) the conductive carbon shell for high electron mobility working in concert. The maximum power density of 16 528 W kg-1 и energy density of 63 W h kg-1 were achieved at around 5 W h kg-1 densities within the voltage range of 1.0-3.5 V of the HSC using the T-Nb2O5@C anode and MSP-20 cathode. The authors claim that these are impressive results.

2.2 Core-shell Hybrid Nanostructures in hybrid supercapacitors

Core-shell nanostructures combine the properties of both components by encircling a core material ($\{TiO2 \text{ in } TiO_2@Ni(OH)_2\}$) with a shell material (Ni(OH)_2 in TiO_2@Ni(OH)_2). The schematic illustration of the core-shell hybrid nanostructure is shown in Figure 3. Better charge transportation is the core's function in addition to supplying electrocapacitance. Metals, semiconductors, and carbon-based

materials make up the three kinds of core materials. Metals and materials based on carbon have higher conductivities than semiconductors. The main function of the carbon and metal cores is to promote charge transmission during the charging and discharging cycles [31]. There are two different configurations in the shell that is growing on the core Making an odd structure, such a layer of nanosheets developing on top of a core of nanowires, is one method. A thin layer forming on top of the core would be an alternative, in which case the morphology of the core would match the structure of the coreshell. The shell's main purposes are to supply a lot of active sites for Faradic redox reactions and a big surface area for charge accumulation [32]. Better charge and ion diffusion are made possible by this combination, which improves electrochemical performance [33].

Core-shell hybrid nanostructures can be fabricated using a variety of materials and techniques and employed in hybrid supercapacitors. For example, Qingqing Ke et al. (2015)[34] synthesize self-supported TiO2@Ni(OH)2 core-shell nanowire arrays on carbon fiber paper by combining hydrothermal synthesis with chemical bath deposition.



Fig 3 ; (a) Schematic depiction of the core/shell system cross section with the skin depth and x-axis direction perpendicular to each other.(b) The average skin depth for the entire region in the computation, as well as the skin depths for Au, Ag, Cu, and Ni [35].

2.3 Classes of core-shell hybrid nanostructures

Core-shell hybrid nanostructures fall into various groups based on their structure and content. There are several main types of core-shell hybrid nanostructures: Single Core/Shell, Multi-Core/Single Shell, Single Core/Multi-Shell, and Multi-Core/Multi-Shell. The categorization of core-shell hybrid nanostructures is displayed in Figures 4 and 5.



Figure 4; Schematic diagram of different forms of core-shell structure [36]. Reprinted with permission from Ref [36].Copyright @ 1920, Copyright Royal Society of Chemisty .



Figure 5: Classification of core-shell nanostructures according to the number of core and shell [37]

2.3.1 Single Core/ single Shell hybrid nanostructure

In order to improve the performance of supercapacitors, a single core/shell hybrid nanostructure for hybrid supercapacitors usually consists of a core material encircled by a shell substance [31]. These structures are perfect for energy storage applications because of their outstanding rate capability, long-term cycling stability, and high specific capacitance [38]. The following are some examples of single core/shell hybrid nanostructures for hybrid supercapacitors: 3D TiO₂@Ni(OH)₂ Core-shell Arrays, in which TiO₂ is the core material and Ni(OH)₂ is the shell material [34], Binderfree NiO/CuO Hybrid Structure, in which NiO is the core and CuO is the shell [39], MnCo₂O₄@Ni(OH)₂ Core-Shell Flowers [40], Nickel Oxide (NiO) Hybrid Core-Shell Nanostructure [41], and NiMoO4@MoS2 Core-Shell Nanorods [42]. The development of a core-shell nanostructure, comprising NiO nanoparticles decorated on polypyrrole nanotube (PNT) via a chitosan (CS) layer (NiO@CS-PNT), is described by Vijeth et al. (2020) [41] as the supercapacitor electrode material. The authors state that a self-degradable soft-template method is used to manufacture PNT. The surface area of polypyrrole (PPy) is increased by the one-dimensional (1D) nanotube framework. By reducing the unavoidable agglomeration of the NiO nanoparticles, CS increases the active material's surface area and improves electrochemical performance.

The NiO/CS-PNT core-shell nanostructure exhibits favorable characteristics such as a high specific capacitance (Csp), low charge transfer resistance (Rct), and a wide surface area when compared to NiO/PNT and pure PNT. An all-solid-state symmetric supercapacitor (SSC) with a high power density (PD) of 4045.69 Wkg-1 and an energy density (ED) of 27.80 Wh Kg-1 was also made using NiO/CS-PNT. Moreover, remarkable cyclic stability was observed, with 84.90% of the capacitance remaining after 10,000 cycles. According to the results, the NiO/CS-PNT core-shell nanostructure makes an excellent electrode material for supercapacitorsThe synthesis of Ni(OH)2@PPy-NTs core-shell nanostructures, which could be employed as supercapacitor electrode materials, is explained by Franciele et al. (2016) [43]. The authors assert that this special Ni(OH)2@PPy-NTs nanoarchitecture improves the electrode's overall electrochemical performance due to the higher electronic/ionic conductivity of 1D nanotubes and the huge surface area provided by 3D nanoflakes. Ni(OH)2@PPy-NTs showed good capacity retention even after 1000 cycles of charge and discharge, with a maximum specific capacitance of 536 F g~1. Not to add, the

low electrochemical series resistance of Ni(OH)2@PPy-NTs was shown by the EIS technique, indicating their potential as a high-performance energy storage medium. In an attempt to enhance supercapacitors' performance, the single core/shell hybrid nanostructure has a core-shell design with programmable nanostructure.

2.3.2 Multi-Core/Single Shell hybrid nanostructure

A multi-core/single shell hybrid nanostructure is a nanoscale material consisting of numerous cores encased in a single shell [44]. These structures are designed to make use of the unique properties of each component to achieve specific goals or functions. Combining numerous cores with a single shell improves the performance of energy storage devices like hybrid supercapacitors [45, 46]. Hybrid nanostructures with several cores and a single shell are engineered to meet certain quality requirements [47]. For example, the inorganic nanocores of the structure can be tailored to promote charge mobility. The performance of the hybrid nanostructure is improved overall by the single shell that surrounds the cores, acting as both a protective and functional layer. One of the many advantages of these hybrid nanostructures is their ability to combine the functions of many elements, leading to improved overall performance and synergistic effects. Furthermore, precise control over the composition and structure of the shell and cores makes customization according to the intended application conceivable [48, 49]. Ni-Co LDH@rGO, where Ni-CoLDH act as the core materials and rGO is the shell material [50], and NiCo2S4@Ni(OH)2/C, where NiCo2S4@Ni(OH)2 is the multi-core and C is the single shell [51], are two examples of multi-core/single-shell hybrid nanostructures in hybrid supercapacitors. Effectively manufactured sandwich-type NiMn2O4@N-C@MnO2 core-shell nanostructures were reported by Chunli Guo et al. (2023) [52]. Microvillous nitrogen-doped carbon (Nsingle bondC), according to the authors, is a highly conductive material that functions as a buffer during repeated charge-discharge cycles, improves electron transport, and increases core-shell nanomaterial conductivity. These properties all contribute to the overall stability of the nanomaterials. The number of active sites in a redox process may rise to some degree with the addition of N. Among all the electrodes that were made, NiMn2O4@N-C@MnO2-2 worked the best. At 10 mA cm-2, the NiMn2O4@N-C@MnO2-2//AC BSH device achieved an energy density of 34.29 W h kg-1, with a power density of 946.75 W kg-1. Moreover, the device retained 96.68% of its initial specific capacitance after 30,000 charge-discharge cycles, which was superior to the bulk of previous published studies on the subject. Because of the device's remarkable electrochemical performance, sandwich-type NiMn2O4@N-C@MnO2 core-shell nanomaterials appear to offer a wide range of potential applications in the development of energyconversion and storage tools. Xin Jiao et al. (2023)[53] produced Fe3O4@onion-like carbons (OLCs) with fewshelled graphitic coatings that were covalently linked. This was achieved in a size-controllable way using monodispersed Fe3O4@oleic acid ligand nanoparticles that were uniform in size. The study, according to the authors, provides a novel method for creating high-performance Fe3O4/carbon hybrid

structures for use in asymmetric supercapacitors. Compared to the majority of known Fe3O4-based electrode materials, the Fe3O4@OLCs hybrid structure demonstrates a high specific capacitance of 686.1 F g-1 at 1 A g-1 and a high rate capability with 71.3% capacitance retention at 10 A g-1. An asymmetric supercapacitor device built using Fe3O4@ OLCs demonstrates a high energy density of up to 63.1 Wh kg-1, which is sustained at a high power density of 1.49 kW kg-1. The energy density is still as high as 39.1 Wh kg-1. Furthermore, a high cycling stability is achieved, with over 80% of the initial capacitance remaining after 10,000 cycles. Investigations on the multi-core/single-shell hvbrid nanostructures' multi-reactive locations indicate that they may play many roles in hybrid supercapacitors. The use of multicore/single-shell hybrid nanostructures to hybrid supercapacitors is therefore an exciting area of research with the potential to significantly increase the versatility and efficiency of energy storage devices.

2.3.3 Single-Core/Multi-Shell hybrid nanostructure

A single-core/multi-shell hybrid nanostructure is a nanoscale material with a single core, usually composed of a particular material encircled by multiple shells made of different materials. These hybrid nanostructures are quite interesting because of their unique properties and applications in various industries, including electronics and energy storage, among others [54]. A nanostructure having a single core at its center encircled by multiple layers or shells composed of different materials is referred to as "single-core/multi-shell" nanostructure. This design allows for the unique properties of each material to be blended, improving performance and functionality. Hierarchical NiCo2S4@ Ni (OH) 2 Core-Shell Hybrid [56], NiO adorned CNT/ZnO Core-Shell Hybrid [55], and SiC@Co(OH)2/Ni3S2 [56] are a few examples of singlecore/multi-shell hybrid nanostructures for hybrid supercapacitors.

Songmin Zhang et al. (2018) [57] created hollow core-shell nanostructures with porous carbon shells encasing T-Nb₂O₅ nanoparticles. The scientists stated that this construction will enhance the high-rate Li-insertion/extraction capabilities of the nanoparticles. Different silica template precursors can be used to create carbon shells with mesoporous channels or relatively dense structures; both T-Nb2O5@carbon hollow core-shell nanostructures show a significant improvement in rate capability over T-Nb₂O₅ nanoparticles. In addition, carbon shells with accessible mesopores of around 6 nm may offer a greater specific surface area and bigger Li+ diffusion channels. Therefore, with a capacity of 173 C g^{-1} at a current density of 50 A g^{-1} , the T-Nb₂O₅@mesoporous carbon (Nb₂O₅@MC) hollow core-shell nanostructures show improved high-rate properties. A range of power delivery applications can benefit from the combination of the Nb₂O₅@MC//MCHS hybrid supercapacitor and mesoporous carbon hollow nanospheres (MCHS) cathode, as it can produce an energy density of 12 Wh kg⁻¹ and an ultrahigh power density of roughly 16 kW kg-1.



Figure 6: schematic representation of Hollow core-shell nanostructures with porous carbon shells enclosing T-Nb₂O₅ nanoparticles [57]

Miao Du et al. (2023)[58] synthesize self-supported (Ni,Co)Se2/(Ni,Co)Se2 composites on carbon fabric with a gourd-like simple approach. The generated sugar (Ni,Co)Se2/(Ni,Co)Se2/CC nanoarrays preserve 88.9% of their capacity after 5000 cycles, achieving a high areal specific capacity of 1.278 mAh/cm2 (220.34 mAh/g) at 2 mA/cm2, according to the authors. Furthermore, the proposed nanocomposite-based flexible hybrid supercapacitor shows remarkable cycle stability and a high energy density of 0.7051 mWh cm-2 at a power density of 1.6303 mW cm-2. These extraordinary energy storage qualities provide strong possibility evidence for the of the **obtained** (Ni,Co)Se2/(Ni,Co)Se2/CC electrode as a battery-type electrode material for supercapacitor applications.

The application of single-core and multiple-shell hybrid nanostructures in hybrid supercapacitors demonstrates the variety of materials and configurations under investigation to increase the supercapacitor technology's efficiency and potential.

3.1.4 Multi-Core/Multi-Shell hybrid nanostructures

An exciting and cutting edge area of study in energy storage is multi-core/multi-shell hybrid nanostructure research, particularly for the creation of hybrid supercapacitors. These nanostructures show promise for next-generation energy storage devices due to their large specific surface area, conductivity, enhanced electrical and enhanced electrochemical performance [59, 60]. Multi-core/multi-shell hybrid nanostructures are designed with numerous cores and shells built of different materials integrated to maximize the overall performance of the supercapacitor [61]. The combination of these materials at the nanoscale produces synergistic effects that increase the capacity, stability, and efficiency of energy storage [62, 63]. Since carbon-based materials have good mechanical strength and electrical conductivity, using graphene, carbon nanotubes, or carbon nanofibers for the core of a hybrid supercapacitor is a popular way to create multi-core/multi-shell hybrid nanostructures.

 Table 2: some reported core-shell hybrid nanostructures and their

 performance in hybridsuperpaganetitors

	perform	nance in hyb	ridsuperp	acapcitors		
Core- shell hybrid nanostru ctures type	Core- shell hybrid nanastru ctures	Specific capacita nce (Fg ⁻¹)	Ener gy densi ty (Whk g ⁻¹)	Power density (Wkg ⁻¹)	Cycl ing stabi lity (%)	Refer ence
Single Core/shel	NiCo ₂ O ₄ /NiMoO ₄	2806	64.2	750	87.7	75
l Single Core/shel	MnO4/N- APC	330	28	560	-	76
Single Core/ shell	TiO ₂ /Ni(OH) ₂	181	53.54	77.71	70	77
Single Core/shel	Fe- Ni/Fe ₂ O ₃ - NiO	1415	27.6	10300	95	78
Singlecor e /multishel	Fe-Ni /Fe ₂ O ₃ @ NiO-HNS	1415.09	11.5	10300	95	79
Single Core/	NiO/CS- PNT	337.11	27.80	4045	69	41
Singlecor e/ Multishel	HCT- 2/Co ₃ O ₄ @SnS ₂	439	40.22	750.22	-	80
Multicore /singleshe ll hybrid nanostruc	PPy@NP CNFs /MnO ₂	595.77	9.36	-	96.2	81
Multicore /singlshel l hybrid nanostruc ture	Ni@CoL DH /rGO	1093	35	3760	-	42
Multicore /singleshe ll	Fe ₂ O ₃ @G @Zn-Ni- Co oxide/	87.5	70	1512	-	82
Multicore /singleshe ll	N1M0O4 C@NiCo2 S4 /Ni(OH)2	-	53.3	6420	98.8	51
Multicore /multishel l	GO@Sn O ₂ /quantum	928.56	25.6	4098	-	83
Multicore /multishel l	Got@ppy FeP-rGO @Ni/	202.1	63.2	3500	-	84
Multicore /multishel l	CuGa ₂ O ₄ @Ni Ppy@CN T- IL/AuNP @CF	38.49 mF cm ⁻²	$\begin{array}{c} 24.7\mu \\ Wh \\ cm^{-2} \end{array}$	$3.52 \text{ m} \\ \text{W cm}^{-2}$	-	85

These carbon cores are subsequently coated with metal oxides, conductive polymers, or other pseudocapacitive materials to form the shell structure [64, 65, 66].

The multi-core/multi-shell configuration of hybrid supercapacitors offers several noteworthy advantages. The electrochemical performance of the supercapacitor can be improved and ion transport accelerated by combining different materials in the core and shell designs [67, 68]. The

synergistic effects of combining diverse materials also lead to improvements in electron transfer kinetics, cycling stability, and charge storage capacity, among other aspects of overall performance [69, 70]. By carefully selecting the materials for the shell and core components, one may tailor the properties of hybrid nanostructures to meet specific objectives, such as high power density, high energy density, or long cycle life [71, 72, 73]. Multi-core/multi-shell hybrid nanostructures' scalable manufacturing technique and adaptable material selection options allow them to be tuned for a variety of applications performance and metrics. Using а straightforward hydrothermal process, Jing Pan et al. (2022) [74] created a hybrid device with a core-shell nanostructure of NiCo2S4@NiV-LDH/NF-2//BC composite material. With a specific capacity of 1778.8 C g^{-1} (3557.6 F g^{-1}) at 1 A g^{-1} , power density of 749.98 W kg-1, and energy density of 120.81 Wh kg-1, the device exhibits exceptional performance. Table 2 shows some reported core-shell hybrid Nanostructures and their performance hybrid in supercapacitors

2.4 Types of core-shell hybrid nanostructure 2.4.1 Yolk-shell hybrid nanostructure

The result of hybridizing hollow and core-shell structures is a yolk-shell structure with a core@void@-shell configuration. Yolk-shell nanostructures have effective properties such as interstitial hollow spaces, movable cores, and shell functioning, making them valuable for a range of applications [86]. The basic procedure for generating yolk-shell nanostructures is the template-assisted technique, which comprises calcination or solvent encapsulation of the core with dual-shell coatings of different materials to remove the inner shell [87, 88]. It has been reported that the Kirkendall effect, also called the reaction with shell [89, 90], together with a few other techniques [91], can be used to produce yolk-shell structures from core-shell nanoparticles.

Silicon is often used as a disposable layer that is removed to leave a vacancy in yolk-shell structures based on magnetic iron oxide NP (MION) [92, 93, 94].

Hydrolysis-controlling agent L-arginine is employed in the hydrothermal process to produce FeOx@SiO2@C NPs. A silica layer and a polydopamine coating were applied to the a-Fe2O3 NPs simultaneously [95]. Analogously, monodisperse and polyethyleneimine (PEI)-modified yolk-shell nanoparticles were used to create Fe3O4@fmSiO2 NPs utilizing an easy-to-follow procedure (Fig. 6). Zhang and others (2012) [96] Interestingly, the internal functionalities of these foreign species have been enhanced by the range of traits bestowed upon them by encapsulation within hollow colloidal particles. To offer hollow particles new properties, researchers have looked into the idea of employing NPs as movable cores [87].

The qualities of core-shell nanoparticles are said to be influenced by their size, composition of material, and coreshell sequence. A hybrid yolk-shell structure furthermore has a movable core housed inside its hollow shell. There is a coresacrificial shell and an outer shell that make up the structure. When the sacrificial shell is removed, a void is left behind that gives the structure its unique properties and multifunctionality.



Figure 7: Transmission electron microscopy pictures of yolk/shell structures encasing various NP core types: a) 90 nm SiO₂ b) 220 nm SiO₂ NPs c) 10 nm Au NPs d) Spindle-shaped Fe₃O₄ nanostructures [97].

2.4.2 Monodispersed core-shell nanoparticles

The fabrication of monodisperse core-shell nanostructures with magnetic moments that is equivalent to 200emu/g is made possible by a unique method. However, within the range of 2-100 nm, size is controlled by adjusting growth factors. A novel structure was created by Velikov et al. that had a silica (SiO_2) shell and a zinc sulfide core, or vice versa. The process of implanted growth was employed to create uniform, well-defined-thickness shells for monodisperse seeds [98]. Cu and Cu@Ag monodisperse nanostructures were produced in a different work. Nevertheless, during synthesizes, three distinct growth mechanisms are at work: metal-assisted growth, galvanic displacement at low temperatures, and overgrowth at high temperatures. Monodisperse core-shell nanostructures can be created using colloidal synthesis [99, 100]. To provide a passivation against Cu oxidation, Cu core Ag shell or Cu@Ag colloids have been created (Fig. 8) [101].



Figure 8: shows TEM pictures of Cu@Ag 4/1 (Cu4@Ag1 Shell produced at T = 25 _C) NPs with HRTEM of polycrystalline Cu@Ag1 NPs and chosen region diffraction bs [101].

2.4.3 Core-shell base on silica nanostructures

Silica is an inorganic material with a variety of applications. Studies have long been conducted on a variety of silica properties, such as its great biocompatibility, stability in acidic environments[102, 103], permeability, etc. [104]. Using the Stober method, tetraethoxysilane (TEOS) is hydrolyzed under simple conditions to create silica nanoparticles (NPs) [105]. Vinyl groups (-C=C) are added to the silica core in the conventional approach of performance enhancement. This successfully promotes the selective adsorption of target molecules. Due to their distinctive properties, silica metal nanostructures—which are composed of silica cores and metal shells—have recently attracted attention. Furthermore, controlled qualities can be produced by altering the morphology, material, and shape of their shells. It is also possible to contemplate using other metals, such as copper, palladium, nickel, and platinum, as nanoshells in silica metal nanostructures. One such metal is gold. The creation of silica metal nanoparticles through seed-mediated growth provides evidence for the control of surface properties and nanoshell thickness.[106].

The silica NPs are coated thinly with conducting polymers to alter their surface via optical, magnetic, optoelectronic, and catalytic properties. Thus, in these core-shell structures, the conducting polymer shells and the very colloidally durable silica cores show similar features. Due to its low cost, ease of synthesis, chemical and ecological durability, high conductivity, and impressive redox activity, polyaniline (PANI) is the most promising technologically advanced conducting polymer[107, 108].

 SiO_2 offers a wide range of advantageous optical applications due to its insulating properties and resistance to high temperatures and liquids. It can be described as "transparent to electromagnetic radiation in the 300–800 nm wavelength range."

On the other hand, silica reacts with boiling hydrogen fluoride or its aqueous solutions, melting Na_2CO_3 and K_2CO_3 , concentrated KOH and NAOH, and silica. Hexafluoro silicic acid can be produced through the reaction of hydrofluoric acid (HF) with silica [109]. Not to mention, silica resists coagulation and is neutral in surface redox reactions [110, 111].

2.4.4 Nanoparticles with a magnetic core-shell

In addition to being magnetic, magnetic core-shell nanostructures have special characteristics. Although there are many uses for iron, cobalt, nickel, and their oxidizing materials or alloys, nickel and cobalt are only utilized in a few particular industries [112, 113,114]. Fe3O4 has so piqued interest among these magnetic materials due to its low toxicity, ease of synthesis, and affordability. The separation of magnetic core-shell nanoparticles by an external magnet is an additional advantage over conventional filtering and centrifugation techniques. In light of this, magnetic core-shell nanoparticles have become an attractive material [115]. Wustite/Fe3O4 core-shell nanocubes are synthesized by Pichon and colleagues. Ferrimagnetic Fe3O4 shell coating antiferromagnetic wustite core causes the hysteresis loop to shift below Neel temperature and enhances coercivity, according to Pichon et al. (2011)[116] magnetic measurement. Saturation magnetization is increased by coating, according to research on core-shell NPs employing $Fe_3O_4@c-Fe_2O_3$ [117] and $a-Fe_2O_3@Fe_3O_4$ [118]. Using

 Fe_3O_4 core-shell nanotubes with a MnZn ferrite shell, magnetic properties are demonstrated [119].

2.4.5 Core-shell polymer nanoparticles

Organizing polymeric shell assembly on silicon and gold nanoparticles has been demonstrated in recent work [120, 121]. There exist multiple approaches for depositing magnetite nanoparticles with a controlled thickness of the polymer shell. Particle size is still greater than 130 nm even if there has been development in the "emulsion polymerization of poly (methyl methacrylate) (PMMA) on 10 nm core of mixed-phase iron oxides" [122, 123]. Block co-polymers distributed the oleic acid-coated superparaMIONs in an aqueous medium using iron oxide NPs anchoring block, such as poly (4-n-dodecyl oxybenzalacetal) [124, 125], poly (styrene250-blockacrylic acid13) [126, 127], or PEG-poly (ecaprolactone) [128].

2.4.6 Nanoparticles with a bimetallic core and shell

Novel and state-of-the-art methods have been devised to control the nanoparticle production process. Since ligands prevent particle aggregation, including them during synthesis is the most widely used method for stabilizing particles. Supramolecular contact governs particle stability and assembly, but covalent and metallic bonding also occur (Fig. 9). Serpell et al. (2011) [129] synthesized gold/palladium coreshell nanoparticles via a supramolecular method. Before reduction, they had already created the metal core surface and had bonded the shell metal to it. It has been shown that the produced core-shell NPs may selectively convert chloronitro benzene into the chemical chloroanaline, which is useful in commerce [129]. Gold/palladium bimetallic nanoparticles were created in another study by using the cavitation phenomenon brought on by high-intensity ultrasonic irradiation exposed to the palladium ions and gold aqueous solution. The produced NPs have a palladium shell surrounding a gold core. Adding sodium dodecyl sulphate to the sample solution causes it to act as a stabilizer and reducer. Mizukashi et al. (2000)[130] state that the thickness of the shell and core is indicated by the concentration of both noble metal ions. Using a one-step seeding-growth process, magnetically recyclable Au@Co core-shell nanoparticles were produced in a matter of minutes [131]. In a different study, Zheng et al. (2006)[132] created Au-Pt core-shell NPs using a displacement reaction, avoiding the production of monometallic AuNPs.



Figure 9: shows a diagrammatic representation of the synthetic anion coordination method for core@shell bimetalllic nanoparticles [129]..

3. Theories/Calculation

Research studies on the structural stability of core-shell hybrid nanostructures for hybrid supercapacitor applications have employed various calculations to evaluate the materials' properties. Some of the calculations used from our findings are presented in the subsection

3.1 Density Functional Theory (DFT) Calculations

DFT calculations are commonly used to investigate the electronic structure, energetics, and stability of materials. Researchers can perform DFT calculations to determine the binding energies, formation energies, and reaction energies of the core-shell hybrid nanostructures. These calculations provide insights into the stability and bonding characteristics of the materials. DFT are supported by some theories which include:

3.1.1 Hohenberg-Kohn Theorems:

The Hohenberg-Kohn theorems form the foundation of DFT. These theorems establish that the electron density uniquely determines the ground-state energy and potential of a system. They provide the theoretical basis for using the electron density as the fundamental variable in DFT calculations. The theorems are stated in equation (1) and (2)

 $E_{HK}(n) = T[n(r)] + V_{int}[n(r)] + \int n(r) v_{ext}(r) d^3r$ (1) $E = E_{HK}, \text{ the exact energy of the system}$ T[n]

where V_{int} is the internal potential for electron-electron interaction,

T(n) is the energy of electron that are in motion,

and $v_{ext}(r)$ is the outside potential at position r for the function for nuclear-electron iteration

$$\langle \Psi' | H | \Psi' \rangle = E' + \int [V_{ext}(r) - V'_{ext}(r)] n_0(r) d^3r$$
 (2)

3.1.2 Kohn-Sham Equations:

The Kohn-Sham equations are a set of equations derived from the Hohenberg-Kohn theorems. These equations reformulate the many-body problem of interacting electrons into an equivalent single-particle problem. Solving the Kohn-Sham equations allows for the determination of the electron density and the total energy of the system. The Kohn-Sham equation is presented in equation 3

$$\left(-\frac{1}{2}\nabla^{2} + V_{\text{eff}}(\mathbf{r})\right)\psi_{i} = \varepsilon_{i}\psi_{i}(\mathbf{r})$$
(3)

Where the effective potential
$$Veff[n(r)]$$
 is given by

$$V_{eff}[n(\mathbf{r})] = V_{ext}(\mathbf{r}) + V_{Hartree}(\mathbf{r}) + V_{EXC}(\mathbf{r})$$
(4)

3.1.3 Exchange-Correlation Functional

The exchange-correlation functional is a key component of DFT calculations. It accounts for the effects of electron exchange and electron correlation, which are crucial for accurately describing the electronic structure and energetics of materials. Various approximations and functionals have been developed to approximate the exchange-correlation energy.

3.1.4 Local Density Approximation (LDA)

LDA is an approximation to the exchange-correlation functional that assumes the exchange-correlation energy depends only on the local electron density at each point in space. LDA provides a computationally efficient approach for DFT calculations but may not capture certain non-local effects accurately.

3.1.5 Generalized Gradient Approximation (GGA):

GGA is an extension of LDA that includes the gradient of the electron density in the exchange-correlation functional. GGA improves upon LDA by incorporating additional information about the electron density distribution, leading to more accurate results for many systems.

3.1.6 Hybrid Functionals

Hybrid functionals combine the exchange-correlation functionals of DFT with a fraction of exact exchange from Hartree-Fock theory. These functionals provide a balance between accuracy and computational efficiency, making them suitable for a wide range of materials and properties.

These theories and approximations support the use of DFT in materials science and quantum chemistry, enabling researchers to calculate electronic structures, energetics, and other properties of materials accurately and efficiently.

3.2 Molecular Dynamics (MD) Simulations

MD simulations are employed to study the dynamic behavior and stability of materials at finite temperatures. Researchers can use MD simulations to analyze the structural changes, thermal stability, and mechanical properties of core-shell hybrid nanostructures. These simulations provide information about the materials' stability under different environmental conditions.

3.3 Monte Carlo Simulations: Monte Carlo simulations are utilized to explore the thermodynamic properties and phase transitions of core-shell hybrid nanostructures. Researchers can use Monte Carlo simulations to investigate the stability of the system, including the formation of defects, phase separation, and ordering of the materials.

3.4 Finite Element Analysis (FEA)

FEA is a numerical method used to analyze the mechanical behavior and structural stability of materials. Researchers can apply FEA to study the stress distribution, deformation, and failure mechanisms of core-shell hybrid nanostructures under different loading conditions. FEA helps assess the mechanical robustness and reliability of the materials for practical applications.

These calculations and simulations enable researchers to comprehensively evaluate the structural stability of core-shell hybrid nanostructures for hybrid supercapacitor applications. By analyzing the materials' properties and behavior, researchers can optimize the design and performance of these nanostructures for efficient energy storage.

4. Methodology

Systematic review method was applied in this paper because it involved a systematic and a transparent approach in order to identify, select, appraise, and analyze relevant studies so that evidence-base conclusion can be provided. A systematic

research is conducted to identify all relevant studies that address the research objectives. Therefore, this paper was conducted across multiple database, providing problems associated with core-shell hybrid nanortsures when put into use in hybrid supercapacitors. solutions were also provided from the findings and a coherent conclusion were drawn from the fingings.

5. Structural Stability Challenges of Core-Shell Hybrid Nanostructures in Hybrid Supercapacitors

5.1 Interface stability challenges

Making sure the core-shell interface is stable is one of the most important factors to take into account when using coreshell hybrid nanostructures in hybrid supercapacitors. Geometric structure, atomic disorder, and surface stress are some of the aspects that affect the interface problems of coreshell hybrid nanostructures [133]. To preserve the structural integrity of the nanostructure, the interface between the materials that make up the core and shell should be robust and resistant to deterioration. Preserving structural stability and averting delamination or shell material separation from the core necessitates optimal interface compatibility between the core and shell materials. A range of tactics have been utilized to improve the stability of the core-shell interface, including surface modification techniques, utilizing appropriate materials with compactible qualities, and fine-tuning the synthesis conditions [134]. The instability of core-shell nanostructures is attributed to surface stress and the interplay between chemical ordering pattern and geometric structure, as noted by Schmidt et al. (2008) [135] and Bochichio et al. (2013) [136]. According to Mangel et al. (2016) [137], atomic disorder at the core-shell interface may be a more important contributor in interface instability than lattice mismatch, which is the conventional theory. In the synthesis of coreshell nanostructures, Wang et al. (2013) [138] highlight the significance of decreasing core-shell interfacial tension and the possibility of utilizing core-shell synergy to address interface instability issues.

5,2 Mechanical stability challenges

The mechanical toughness of the core-shell hybrid nanostructures is an additional factor contributing to their structural stability. When various materials are integrated into core-shell systems, mechanical stress and strain may result from these cycles of charge and discharge. This could eventually lead to structural deterioration and decreased performance. For a nanostructure to function well in applications involving mechanical stress or strain, its mechanical characteristics—such as its strength, flexibility, and resistance to deformation are essential [138]. In order to increase the mechanical stability of core-shell hybrid nanostructures, researchers have looked into a variety of strategies, including creating hierarchical structures, adding reinforcing materials, and enhancing the composition and shape of the nanostructure [139,140]. When it comes to the mechanical stability of core-shell hybrid nanostructures in hybrid supercapacitors, the flexibility attribute is essential.

nanostructures are crucial parts of flexible These supercapacitors since they are made to perform both mechanically and electrochemically [141]. The entire flexibility and durability of the supercapacitor device depend on the mechanical flexibility of these core-shell hybrid nanostructures. These nanostructures' flexibility helps the supercapacitor maintain its electrochemical performance by allowing it to adjust to various mechanical circumstances without losing functionality [142]. Flexibility is an essential component of mechanical stability, as was previously mentioned, when designing core-shell hybrid nanostructures for usage in hybrid supercapacitors. This property is important, as shown by the studies of Qu et al. (2020) [143] and Sekhar et al. (2019) [135], who show the excellent electrochemical performance and stability of moss-like nickel-cobalt phosphide nanostructures, and Sekhar et al. (2019) [135] who show the high capacity and long-term cycling durability of core-shell-like copper hydroxide nanotube arrays grafted with nickel aluminum layered double hydroxide nanosheets. Veerasubramani et al. (2017) [144], who used a binderless cobalt hydroxide-cobalt molybdate core-shell structure, further highlight the importance of flexibility in liquid electrolyte mediated flexible pouch-type hybrid supercapacitors. Last but not least, Zheng et al. (2013) [145] report TiO2@C core-shell nanowires for flexible and high-performing solid-state supercapacitors, highlighting the significance of flexibility for these nanostructures' mechanical stability.

Strength is the nanostructures' capacity to retain their structural integrity and fend off deformation or failure in the face of mechanical stress or strain in the context of the mechanical stability of core-shell hybrid nanostructures in hybrid supercapacitors. These nanostructures' mechanical strength plays a critical role in how well supercapacitors function since it directly affects their long-term stability and endurance, particularly in applications that need frequent cycles of charging and discharging. Research by Liang et al. (2018) [146] and Sekhar et al. (2019) [147] emphasizes how crucial this feature is to obtaining high capacity and long-term cycle durability. Superior cycle stability is demonstrated by the crystalline/amorphous Co3S4 core-shell heterostructure in Liang's work and the core-shell-like nanoarchitectures in Sekhar et al.'s work, highlighting the importance of mechanical stability in their functionality. The importance of core-shell nanostructures in enhancing structural mechanical stability is further emphasized by Jiang et al. (2016) [45], as this is crucial for the long-term performance of supercapacitors. Finally, Wang et al. (2017) [148] emphasize that the mechanical stability of the core-shell FeOF/Ni(OH)2 hierarchical nanostructure plays a crucial role in attaining high performance, as do the unique microstructure and synergistic effects. In the context of hybrid supercapacitors, where core-shell nanostructures significantly improve the overall performance and dependability of the energy storage devices, the idea of strength is very crucial.

5.3 Electrochemical stability challenges

The combination of various materials and their features in core-shell hybrid nanostructures creates significant obstacles

for electrochemical stability in hybrid supercapacitors. In hybrid supercapacitors, these difficulties may affect the longevity and functionality of these nanostructures. Material compatibility, oxidation and corrosion, electrochemical reactivity, and structural integrity are the main causes of the electrochemical stability problems with core-shell hybrid nanostructures [149, 150]. Combining several materials, such as metals and biomolecules, so that one serves as the shell and the other as the core, is a common technique used in coreshell hybrid nanostructures. Maintaining stability and preventing degradation can be difficult when these materials are not compatible with the electrochemical environment [151]. A potential field of study is the construction of coreshell hybrid nanostructures for electrochemical applications, emphasis improving stability with an on and performance. According to Jiang et al. (2016) [109] and Feng et al. (2018) [152], these structures have a lot of potential for electrochemical energy storage, especially in Li-ion batteries and supercapacitors. They talk about how different materials can be used, such as metal-phosphide and carbon-based coreshell nanostructures, and they stress how crucial stability and compatibility are to these designs. The usage of polymer nanofibre membranes with a core-shell structure in gel polymer electrolytes is further explored by Bi et al. (2014) [153], who show that these membranes have high ionic conductivities and stability. A particular example of a coreshell nanostructure is the Fe@FeP/CNT material, which shows remarkable catalytic activity in the hydrogen evolution reaction, as reported by Li et al. (2017) [154]. The compatibility of electrolytes and core-shell nanostructures plays a significant role in hybrid supercapacitor design. In order to achieve high ionic conductivities and stability, Bi et al. emphasize the significance of the microstructure and the affinity between the electrolyte and the core-shell structure. Core-epitaxial shell nanocrystals are a unique way of stabilizing battery electrode/electrolyte interfaces that Kim et al. (2015) [155] introduces and that may be able to solve some of the compatibility issues. The cumulative findings of these investigations highlight the promise of core-shell hybrid nanostructures in electrochemical settings as well as the necessity for additional study to maximize their functionality and design. These difficulties demonstrate the necessity of more study and advancement in this field.

When different materials are present in core-shell nanostructures, oxidation and corrosion problems may arise, particularly in electrochemical settings. This results from the asymmetric paths for corrosion as well as the possibility of nanoparticle migration and coalescence during catalytic reactions [156, 157]. Over time, this can have an impact on the nanostructures' performance and stability. On the other hand, these problems can be lessened by using porous shells that let chemical species in freely and regulate how various parts of the nanostructure interact with one another [158]. By combining metallic, porous metal oxide, and carbon nanoparticles into polymeric matrix, hybrid nanostructured coatings can also protect base metals from corrosion. This allows for the creation of multifunctional coatings. [159]

Structural stability in electrochemical processes may be hampered by the reactivity of core-shell materials. For instance, it might be challenging to regulate the development of passivating layers on the electrode surface of Li-ion batteries, which can result in undesirable reactions and decreased durability [160]. Comparably, despite their excellent power delivery and long-term cycling stability, core-shell nanostructures' special qualities in electrochemical capacitors can make it difficult to maintain structural stability [106]. In order to overcome these obstacles, [161] and [162] suggest methods for enhancing the structural stability of coreshell materials. These methods include the use of atomically thin Pt shells to shield the core from oxidation and dissolution and the use of core-shell nanocomposites to counteract negative surface effects and stabilize the materials. The overall stability of the nanostructures can be affected by the interaction between the core and shell materials when electrolytes or other electrochemically active chemicals are present.

The ability of core-shell hybrid nanostructures to preserve physical and chemical integrity during the their supercapacitor's operating conditions is referred to as structural integrity in the context of hybrid supercapacitors. A key component of the performance and longevity of hybrid supercapacitors is the structural integrity of their core-shell hybrid nanostructures. Research has indicated that the electrochemical performance and stability of supercapacitors can be greatly improved by employing core-shell-like nanoarchitectures, such as hierarchical CuCo2S4@NiMnlayered double hydroxide core-shell hybrid arrays [249], hierarchical polyaniline-coated NiCo2S4 nanowires [248], and copper hydroxide nanotube arrays grafted with nickel aluminum layered double hydroxide nanosheets [116]. With their high electron diffusion efficiency, plenty of readily available electroactive sites, and enhanced structural stability, these architectures produce large specific capacitance, strong rate capability, and extended cycle lifetimes.

The performance of supercapacitors may also be enhanced by the application of core-shell nanostructures, such as singlewall carbon nanotubes covered with covalent organic frameworks [162]. To address the particular problems with structural integrity in these hybrid nanostructures, more investigation is necessary.

5.4 Cycling stability challenges

The ability of core-shell hybrid nanostructures to retain their structural and functional integrity across multiple cycles of use, especially in the face of changing environmental conditions or external stimuli, is referred to as cycling stability. This characteristic is essential for the real-world uses of core-shell hybrid nanostructures in hybrid supercapacitors. The long-term performance and dependability of core-shell hybrid nanostructures are significantly influenced by cycling stability [116, 132]. For instance, good cycle stability in hybrid supercapacitors was established by the core-shell-like copper hydroxide nanotube arrays grafted nickel aluminum layered double hydroxide nanosheets in the work of Sekher et al (2019)[116]. Comparably, Liang et al. (2018)[132] revealed that a crystalline/amorphous Co3S4 core-shell heterostructure exhibited higher cycling stability in aqueous hybrid supercapacitors. Excellent cycling performance was

also facilitated by the creation of self-supported yolk-shell nanocolloids with high surface area and structural stability, as reported by Zhu et al. (2015)[163]. It has an immediate effect on how suitable they are for real-world uses, particularly in situations where frequent usage or exposure to various environments is anticipated. The significance of the stabilizing agent in preserving the macroscopic characteristics of hybrid nanomaterials is emphasized by Spataro et al. (2020)[164]. Bowles et al. (1988)[253] and Razak et al. (2019)[165] both highlight how thermal cycling affects the physical properties and dimensional stability of hybrid composites, with Bowles et al. highlighting the enhanced stability of specific hybrid laminates. Yao et al. (2012)[166] highlight the importance of cycling stability in real-world applications by showing how conductive polymer coatings can greatly improve silicon nanowire anode cycling stability. Together, these investigations highlight how important cycling stability is in deciding whether hybrid nanostructures are suitable for real-world applications.

Numerous factors affect the core-shell hybrid nanostructures' cycle stability. The thermal behavior and thermodynamic stability of yolk-shell nanoclusters can be influenced by the size of the shell, core, and void space [156]. Another important factor is where the core is located within the shell; in Ni-Au core-shell nanoparticles, decentralized cores promote greater stability [167]. In addition, the thermoresponsiveness and colloidal stability of nanoparticles can be affected by the kind of polymer topology in the shell, such as cyclic versus linear [168]. These results emphasize how crucial it is to take into account a variety of structural elements when designing and refining core-shell hybrid nanostructures in order to maximize cycling stability.

Core-shell hybrid nanostructures have been shown to have the potential to improve cycling stability in a variety of applications. For example, Wang et al. (2016) [169] designed highly cross-linked Cu/a-Si core-shell nanowires for use in lithium batteries, achieving ultra-long cycle life and high rate performance. Jiang et al. (2016) [109] reviewed the use of core-shell nanostructures in electrochemical capacitors, highlighting their ability to provide high power delivery and long-term cycling stability. Recent research has shown just how beneficial core-shell hybrid nanostructures can be in a variety of applications. Zhu et al. (2015) proposed selfsupported yolk-shell nanocolloids, exhibiting high capacitance and outstanding cycling behavior, for use in pseudocapacitors. All of these investigations highlight how core-shell hybrid nanostructures can improve cycling stability for a variety of energy storage uses.

5.5 Material Degradation:

The structural stability of core-shell hybrid nanostructures in hybrid supercapacitor applications is seriously threatened by material deterioration [170]. By offering a wide surface area and excellent conductivity, core-shell nanostructures are intended to improve the electrocapacitive performance of supercapacitors [171]. Nevertheless, these nanostructures are susceptible to degradation with time, which could impair their overall performance and structural stability [172]. Exposure to extreme working conditions, such as high temperatures, humidity, and chemical reactions, is one of the primary causes of material degradation in core-shell hybrid nanostructures [173]. These circumstances may cause the materials in the core and shell to deteriorate, resulting in structural damage and decreased functionality.

Material deterioration can also be facilitated by the supercapacitor's cycle of charging and discharging [174]. Every cycle puts stress on the core-shell nanostructures, which can result in mechanical fatigue and the development of flaws or fissures. These flaws have the potential to worsen the nanostructures' structural stability and hasten the process of deterioration.

Furthermore, material deterioration may also be facilitated by contaminants or other impurities in the core-shell materials [175]. The materials may react with these contaminants, triggering chemical reactions that hasten the deterioration process. To reduce the chance of deterioration, it is essential to guarantee the purity of the materials employed in core-shell hybrid nanostructures [175].

Researchers are looking into a number of approaches to lessen the difficulties caused by material degradation. To shield the core-shell nanostructures from challenging working circumstances and stop degradation, one strategy is to create protective coatings or encapsulation approaches. Mitigating the degradation of core-shell hybrid nanostructures in hybrid supercapacitors is a critical function of protective coatings. Since the electrodes are the most crucial parts of a supercapacitor cell, these coatings are intended to improve their functionality and longevity of the material during operations. Using protective coatings, a layer of material is applied to the core-shell nanostructures' surface to act as a barrier of defense. The stability and durability of the nanostructures can be improved by designing these coatings with specific properties. These coatings are often made of polymers, epoxies, polyurethanes, and metals like aluminum and zinc. For instance, for all-solid-state flexible supercapacitors, K. Qi et al. (2018)[176] created a core-shell tubular structured graphene nanoflake-coated polypyrrole nanotube (GNF/PNT) hybrid. The GNF coating does not only function as an effective surface protector but aslo an electron transport channel, while a precisely calibrated coating quantity maximizes the composite's overall capacitance, according to the authors. Thus, the GNF/PNT hybrid exhibits a stable cycle behavior, a large capacitance, and a high capacitance retention. Encapsulation techniques entail the encasement of core-shell nanostructures in a matrix or protective shell. The nanostructures are physically protected from outside influences by this encapsulation. For example, a work by Kavinkumar et al. (2021) [177] described a class of electrodes that are core-shell type and are encapsulated in a homogeneous outer nanolayer. With a model electrode consisting of a $NiCo_2O_4/MoO_2$ core-shell encased by an atomic layer-deposited NiO nanolayer, the authors proved the effectiveness of the kind of electrodes. Based on DFT calculations and experimental investigation, the authors assert that the electrical modulation at the interface formed by the outer nanolayer offers a significant increase in capacity more than 450 C g^{-1} for the model electrode. After 20,000 cycles

of charge and discharge for the model electrode, the flexibility provided by the thin nanolayer helps maintain the physical integrity of the encapsulated inner structure, lowering the loss in capacity to less than 3% from more than To increase the stability and endurance of the 10%. nanostructures, another tactic is to optimize the synthesis techniques and material compositions. For example, using a one-step hydrothermal process, Lingli Cheng et al. (2018) [178] effectively designed and synthesized size-t3nable hierarchical hollow core/shell submicrospheres based on nickel sulfide, in which Ni3S2/NiS hollow submicrosphere functions as a core and Ni3S4 nanoflakes as shells. The authors assert that these hybrids have better electrochemical performance because of the complementary effects of different phases of nickel sulfide. To sum up, in hybrid supercapacitor's applications, material degradation presents a major barrier to the structural integrity of core-shell hybrid To solve these issues and enhance the nanostructures. durability and performance of core-shell hvbrid nanostructures in supercapacitors, researchers are currently working on creating safeguards and refining synthesis techniques.

6. Results and Discussions

6.1 Strategies for Enhancing Structural Stability in Core-Shell hybrid Nanostructures

6.1.1 **Protective Coating or Encapsulating Technique**

Protective coating or encapsulating is a method used to address the problem of structural stability in core-shell hybrid nanostructures for hybrid supercapacitor applications [269]. This method involves applying a protective layer or shell around the core of the nanostructure to enhance its stability and prevent degradation or structural damage [179].

The use of protective coatings in hybrid supercapacitors can improve the structural integrity of the core-shell nanostructures, enhance their mechanical strength, and protect them from external factors such as moisture, temperature variations, and chemical reactions [180]. One common material used for encapsulating core-shell nanostructures is silica due to its excellent colloidal stability, controllability in synthesis, chemical inertness, controlled porosity, high processability, and optical transparency [181]. By encapsulating the core nanomaterials with silica, the exterior surface properties can be modified, and colloidal stability can be enhanced [181].

The encapsulation process involves depositing a shell layer, such as silica, onto the core of the nanostructure. This shell layer encapsulates the entire core, providing a protective barrier. The protective coating of core-shell hybrid nanostructures in hybrid supercapacitors is crucial for maintaining their structural stability and ensuring long-term performance. It allows for the efficient storage of electrochemical energy and helps optimize the power and energy density of the supercapacitors. K. Qi et al. (2018)[182] created a core-shell tubular structured graphene nanoflake-coated polypyrrole nanotube (GNF/PNT) hybrid. The GNF coating does not only function as an effective surface

protector but aslo an electron transport channel, while a precisely calibrated coating quantity maximizes the composite's overall capacitance, according to the authors. Thus, the GNF/PNT hybrid exhibits a stable cycle behavior, a large capacitance, and a high capacitance retention. A work by Kavinkumar et al. (2021) [183] described a class of electrodes that are core-shell type and are encapsulated in a homogeneous outer nanolayer. With a model electrode consisting of a NiCo₂O₄/MoO₂ core-shell encased by an atomic layer-deposited NiO nanolayer, the authors proved the effectiveness of the kind of electrodes.

In summary, protective coating is a method used to enhance the structural stability of core-shell hybrid nanostructures in hybrid supercapacitors

6.1..2 Surface Functionalization Techniques

Surface functionalization techniques are used to address the problem of structural instability in core-shell hybrid nanostructures in hybrid supercapacitor applications [184]. Through these methods, the stability and functionality of the nanostructures are improved by altering their surface. Supercapacitors can greatly increase their specific and volumetric storage capacity by carefully controlling the surface characteristics of the electrode materials. Surface functionalization provides more surface area for ion adsorption, increasing the amount of ions that may be stored. It shortens the nanostructures' ion diffusion lengths, facilitating quicker ion transport and improving electrochemical performance all around [183,184]. Some common approaches include: Chemical Modification, Coating, Doping, and Hybridization.

In order to change the surface chemistry and reactivity of core-shell hybrid nanostructures, functional groups or particular molecules are added to their surface [185]. Samia et al.(2011) [185] presented a review on the use of diazonium salts as a new class of surface modifiers and coupling agents for attaching artificial polymers, biomacromolecules, and nanoparticles to surfaces. Surface functionalization can be achieved using a variety of techniques, such as covalent and non-covalent attachment of functional groups or molecules to the surface of the core-shell hybrid nanostructure. Covalent functionalization is the process by which molecules or functional groups attach themselves to the surface of a coreshell hybrid nanostructure by forming strong chemical interactions. Depending on the type of functional groups to be attached, covalent attachment can be accomplished by a variety of chemical processes, including diazonium coupling, thiolation, and silanization [186]. In the case of silanization, for example, the surface hydroxyl groups of nanostructures react with silane coupling agents, resulting in the covalent attachment of functional groups, like aryl or alkyl moieties, to the surface [187]. Zhenghu et al. (2017)[188] described the construction of а one-dimensional coaxial CNT@microporous carbon composite (CNT@micro-C) using surface-confined cross-linking generated by covalent bonds. The precursor for microporous carbon chosen by the authors is octaphenyl polyhedral oligomeric silsesquioxane (Ph-POSS), which is made up of eight phenyls and a $-Si_8O_{12}$ cage. The authors harvested CNT@micro-C, which consists

of a microporous carbon shell and a CNT core, after etching and carbonizing -Si₈O₁₂ cages. This process involved layerby-layer cross-linking of phenyl anchored Ph-POSS on the surface of CNT. According to the authors, the surface area of CNT@micro-C can reach 1306 m² g⁻¹, and the thickness of the microporous carbon shell can be precisely controlled between 6.0 and 20.0 nm. With a wide surface area, high electrical conductivity, quick ion transfer speed, and short ion transfer distance, CNT@micro-C combines the structural benefits of carbon nanotubes and microporous carbon. In a related development, Neha et al.(2023)[189] employed aryl diazonium salts as a fascinating element to alter the surfaces of NiFe₂O₄ nanoparticles. This was achieved through a radical pathway that formed strong covalent links, resulting in stable interfaces at the NP-organic molecules. The authors claim that in order to guarantee NP production and covalent surface functionalization, this complimentary surface analysis is used. Non-covalent surface functionalization is a method for modifying the surfaces of core-shell hybrid nanostructures without forming strong covalent bonds [190]. Using weak intermolecular forces such as hydrogen bonding, π - π interactions, van der Waals forces, electrostatic interactions, and hydrophobic effects, non-covalent functionalization allows molecules to be attached to the surface of the coreshell nanostructures [191]. This technique does not cause any harm to the original structure of the material [192]. This is essential to preserving the nanostructure's stability in the face of the mechanical strain and electrochemical cycling seen in supercapacitor applications. Self-healing properties are made possible by the weak and reversible nature of non-covalent bonding [193]. The functional layer has the ability to selfassemble if it is broken or separated, preserving the structural integrity over time. Doping is a surface functionalization procedure whereby impurities are added to a nanostructure's surface to modify its electrical properties [194]. Through modifications to the surface chemistry or an increase in conductivity, this can improve specific functionality.

6.1.3 Optimization of Synthesis Methods

One effective tactic for improving the structural stability of core-shell hybrid nanostructures is synthesis techniques optimization. By improving the synthesis process, the resulting nanostructures' robustness and durability can be raised. The search results lead to the following conclusions. Optimizing the synthesis process involves controlling the nucleation and growth processes. Temperature, reaction time, and precursor concentrations can all be carefully controlled to gain more control over the nucleation and growth of the core and shell components [195]. By reducing flaws and guaranteeing nanostructure homogeneity, this control can result in increased structural stability. Elena et al. (2003)[195] investigated how CoPt₃ nanocrystal size is controlled by nucleation rate. In order to understand the mechanisms controlling the creation of the metal alloy nanocrystals, many experimental parameters (such as reaction temperature, stabilizing agent concentration, cobalt to platinum precursor ratio, etc.) were examined in their work. The authors demonstrate that CoPt₃ nanocrystals nucleate and develop to their full size early in the synthesis process, and that additional heating does not cause Ostwald ripening. In this

instance, controlling the size of the nanocrystals requires striking the right balance between their rates of nucleation and development from molecular precursors. As a result, the authors concluded that $CoPt_3$ nanocrystals' size can be accurately adjusted in a predictable and repeatable manner between 3 and 18 nm.

Adjusting the ratio and mix of core and shell components is another tactic. The structural stability of the core-shell hybrid nanostructures can be improved by choosing suitable materials with complementary qualities [196].. A thorough understanding of the materials, synthesis methods, and desired properties of the nanostructures is necessary for the difficult process of optimizing the synthesis method for structural stability augmentation. To evaluate the efficacy of the optimization procedures, experimental validation and characterization approaches are necessary.

To sum up, one effective tactic for improving the structural stability of core-shell hybrid nanostructures is synthesis technique optimization. The overall structural integrity and durability of these nanostructures can be enhanced by regulating the nucleation and growth process, adjusting the ratio and composition of core and shell materials, and using post-synthesis treatments.

6.1.4 Optimization of Materials Composition

Optimizing the composition of materials used in the synthesis of core-shell hybrid nanostructures is a valuable strategy for enhancing their structural stability [197]. By carefully selecting and designing the materials, it is possible to improve the overall stability and durability of the nanostructures.

One approach to optimizing materials composition is through multi-material designs [198]. This involves using multiple materials with complementary properties in the core-shell nanostructure [199]. By selecting materials with similar thermal expansion coefficients and mechanical properties, it is possible to reduce stress at the interface and enhance the overall stability of the nanostructures [200]

Another technique is topology optimization, which aims to find the optimal distribution of materials within the nanostructure to maximize stiffness and minimize mass. Topology optimization considers the arrangement of materials at a macroscopic level, allowing for the identification of material layouts that enhance structural stability [201]. This technique can be particularly useful when dealing with complex parts carrying multiple loads, as it helps identify material layouts that maximize stiffness and minimize mass. In addition to topology optimization, size, shape, and freeshape optimization can be employed to fine-tune the formation of structural product concepts. These optimization techniques help identify optimal solutions for key product characteristics such as cross-sectional thickness, material choice, and areas with high stress concentration. By refining these areas, the risk of product failure can be reduced, leading to improved structural stability

Furthermore, introducing vacancy engineering, substitution/gap doping, object modification, and combination engineering can optimize the cycle stability and structural

stability of cathode materials for various applications, such as aqueous zinc ion batteries [202]. These strategies aim to inhibit material dissolution, improve reaction reversibility, and enhance the overall structural stability of the cathode materials.

It is important to note that the optimization of materials composition for structural stability enhancement of core-shell hybrid nanostructures is a complex process that requires a deep understanding of the core-shell materials and their properties. Experimental validation and characterization techniques are essential to assess the effectiveness of the optimization strategies.

In conclusion, optimizing the composition of materials used in the synthesis of core-shell hybrid nanostructures is a valuable strategy for enhancing their structural stability. By carefully selecting materials, employing multi-material designs, utilizing topology optimization, and considering vacancy engineering and doping techniques, it is possible to improve the overall stability and durability of these nanostructures.

6.1.5 Designing Hierarchical Structures

Designing hierarchical structures is a powerful strategy for enhancing the structural stability of core-shell hybrid nanostructures. Hierarchical structures, which incorporate multiple levels of organization, can improve the mechanical properties and overall stability of these nanostructures. By carefully designing the hierarchical architecture, it is possible to distribute stress, enhance load-bearing capacity, and minimize the risk of structural failure. Here, the concept of designing hierarchical structures as a strategy for structural stability enhancement specifically for core-shell hybrid nanostructures is explored.

One approach to designing hierarchical structures is through architectural design. By organizing the core and shell materials in specific patterns or arrangements at different length scales, it is possible to optimize their interaction and enhance structural stability [203]. For example, incorporating a hierarchical arrangement of core-shell layers, with progressively changing shell thicknesses or material compositions, can distribute stress and strain more effectively, improving the overall stability of the nanostructure [204]. <u>Matthew</u> et al.(2016)[203] presented three methods of organizing hierarchical structures which include: ligands placed on the surface of nanostructures, atoms organized into crystalline nanostructures, and nanostrsuctures organized into crystalline superlattices

Another strategy is hierarchical reinforcement [205]. This involves incorporating reinforcing elements at multiple length scales within the core-shell hybrid nanostructure. By introducing nanofillers, such as nanoparticles or nanotubes, into the core or shell materials, it is possible to enhance their mechanical properties and reinforce the structure. The addition of these reinforcements at different length scales can improve strength, toughness, and resistance to deformation, leading to enhanced structural stability. Additionally, self-assembly techniques can be employed to create hierarchical structures within core-shell hybrid nanostructures. Self-assembly processes, such as molecular self-assembly or electrostatic self-assembly, allow the materials to organize themselves into ordered structures spontaneously. By utilizing self-assembly, it is possible to create complex hierarchical architectures with precise control over their organization and arrangement. This can lead to improved load distribution, stress transfer, and overall structural stability of the core-shell hybrid nanostructures. For instance, Qiannam et al.(2015)[206] presented a two-step self-assembly techniques for hierarchical-ordered nanostructures: Two-step self-assembly carrying out two discontinuous processes and two-step self-assembly finished in one pot involving two continuous processes. According to the authors, the two-step self-assembly strategy, in contrast to the traditional one-step method, enables the fusion of several and the creation of intricate synthetic processes hierarchically-ordered multiscale nanostructures with structures. Furthermore, this strategy makes it possible for heterostructures or hybrid nanomaterials to self-assemble economically. It is anticipated that the broad use of two-step self-assembly would open up new avenues for the fabrication multifunctional nanostructures with purposeful of morphologies. Multiple-step self-assembly is the extension of the two-step self-assembly idea to syntheses that involve more than two chemical or physical reaction steps.

Furthermore, surface engineering plays a crucial role in designing hierarchical structures at the nanoscale [207]. By modifying the surface properties of the core-shell nanostructures, such as through nanostructuring or introducing coatings, it is possible to enhance adhesion, reduce stress concentrations, and improve interfacial compatibility between the core and shell materials [208, 209]. These surface modifications can contribute to the overall stability and durability of the nanostructures.

In conclusion, designing hierarchical structures is an effective strategy for enhancing the structural stability of core-shell hybrid nanostructures. This approach allows for the development of more robust and durable core-shell hybrid nanostructures, expanding their potential applications in various fields.

6.1.6 Utilization of Appropriate Materials With Compactible Qualities

Utilizing appropriate materials with compatible qualities is indeed a strategy for enhancing the structural stability of coreshell hybrid nanostructures [210]. By carefully selecting materials that have compatible properties, it is possible to improve the overall stability and durability of the nanostructures. Here are some insights based on the search results:

One important consideration is the lattice matching between the core and shell materials [211]. In some cases, lattice mismatches between the core and shell can lead to structural instabilities [212]. However, Jia & Jiatao (2020)[213] has reported a synthetic route that achieves nanoscale heterostructures with substantial lattice mismatches between the core and monocrystalline semiconductor shell. This is

achieved by controlling soft acid-base coordination reactions between molecular complexes and colloidal nanostructures, allowing for the growth of a semiconductor shell with an incommensurates lattice structure with that of the core.

Another aspect to consider is the compatibility of materials at the interface. When the core and shell materials have different thermal expansion coefficients or mechanical properties, stress concentrations can occur at the interface, leading to structural deformations or failures [214]. By selecting materials with similar properties, such as thermal expansion coefficients and mechanical strength, it is possible to reduce stress at the interface and enhance the overall structural stability of the core-shell hybrid nanostructures.

Furthermore, the choice of materials with appropriate surface properties can also contribute to structural stability [215, 216]. Surface engineering techniques, such as nanostructuring or introducing coatings, can improve adhesion, reduce stress concentrations, and enhance interfacial compatibility between different materials. These surface modifications can enhance the overall stability and durability of the nanostructures [217, 218].

In conclusion, utilizing appropriate materials with compatible qualities is a valuable strategy for enhancing the structural stability of core-shell hybrid nanostructures. By considering factors such as lattice matching, compatibility at the interface, and surface properties, it is possible to improve the overall stability and durability of these nanostructures.

7. Conclusion and Future Scope

An in-depth understanding of the current status of research in this subject has been made possible by this extensive review on the structural stability of core-shell hybrid nanostructures for hybrid supercapacitor applications. This review has consolidated important discoveries and highlighted significant trends and problems in the quest to improve the structural stability of these nanostructures by looking at a wide range of studies.

The research findings demonstrated that core-shell hybrid nanostructures have enormous potential for improving the longevity and efficiency of hybrid supercapacitors. Improved mechanical strength, increased electrical conductivity, and extended cycle life are just a few advantages of the novel design, which combines a stable core material with a functioning shell. These properties are essential to supercapacitors' effective and dependable performance in a variety of applications, such as portable electronics, renewable energy systems, and electric vehicle energy storage.

Furthermore, the thorough analysis has provided significant new information on the factors affecting the structural stability of core-shell hybrid nanostructures. Overall stability and performance of these nanostructures have been found to be significantly influenced by a number of parameters, including the synthesis method, the interface between the shell and core, mechanical toughness of the core-shell materials when in operation, electrochemical reactivity of the core-shell material, cycling stability of the core-shell materials in the face of changing environmental condition, and the choice of materials for the core and shell. Designing and refining core-shell hybrid nanostructures to maximize stability and performance in supercapacitor applications requires an understanding of these variables. Through protective coating or encapsulation, surface functionalization, optimization of synthesis methods, optimization of materials composition, designing hierarchical structures, and utilization of appropriate materials with compactible qualities, researchers have made significant progress in improving the electrochemical performance and durability of these nanostructures

This review has also pointed out a number of issues and directions for further investigation. For example, more research is required to determine the long-term stability of core-shell hybrid nanostructures under various operating conditions, such as mechanical stress, humidity, and temperature fluctuations. To enable their broad use in useful supercapacitor devices, it is also necessary to address the scalability and cost effectiveness of these nanostructures.

This extensive review has shown how core-shell hybrid nanostructures can significantly improve the structural stability of hybrid supercapacitors.

For future research and development projects aiming at enhancing the functionality, durability, and commercial feasibility of these nanostructures, the conclusions and insights offered in this evaluation offer a strong foundation. Core-shell hybrid nanostructures has the potential to transform the field of energy storage and aid in the development of efficient and sustainable energy systems, provided that material synthesis, characterization methods, and device engineering advances further.

Conflict of Interet\st

The author declare that there is no conflict of interest

Funding Source

None

Authors' Contributions

The corresponding author conceive and write every part of the articlr

Acknowledgement

None

Reference

- [1] Mohammad Amir, Radhika G. Deshmukh, Haris M. Khalid, Zafar Said, Ali Raza, S.M. Muyeen, Abdul-Sattar Nizami, Rajvikram Madurai Elavarasan, R. Said, Kamaruzzaman Sopian, "Energy storage technologies: An integrated survey of developments, global economical/environmental effects, optimal scheduling model, and sustainable adaption policies", Journal of Energy Storage 72, Part E, pp.108694, 2023
- [2] P.C. Himadri Reddy, John Amalraj S. Ranganatha, Smitha S. Patil, Saravanan Chandrasekara, "A review on effect of conducting polymers on carbon-based electrode materials for

electrochemical supercapacitors", Synthetic Metals, Vol. 298, pp.117447, 2022

- [3] <u>Dhruba P. Chatterjee</u>, Arun K. Nandi, "A review on the recent advances in hybrid supercapacitors," J. Mater. Chem. A , Vol. 9, pp.15880-15918, 2021
- [4] Elumalai Dhandapani, Sadhasivam Thangarasu , K. Ramesh^c, R. Vasudevan, Navaneethan Duraisamy , "Recent development and prospective of carbonaceous material, conducting polymer and their composite electrode materials for supercapacitor — A review", Journal of Energy Storage, 52, Part C, pp.104937, 2022
- [5] Sultan Ahmed, Ahsan Ahmed, D. Baba Basha, Shahir Hussain, Islam Uddin, M.A. Gondal, "Critical review on recent developments in conducting polymer nanocomposites for supercapacitors", Synthetic Metals, <u>Vol. 295</u>, pp.117326, 2023
- [6] Chen-qi YI, Jian-peng ZOU, Hong-zhi YANG, Xian LENG , "Recent advances in pseudocapacitor electrode materials: Transition metal oxides and nitrides", Transactions of Nonferrous Metals Society of China, Vol. 28, Issue 10, pp. 1980-2001, 2018
- [7] Wei Hau Low, Poi Sim Khiew, Siew Shee Lim, Chiu Wee Siong, Ejikeme Raphael Ezeigwe, "Recent development of mixed transition metal oxide and graphene/mixed transition metal oxide based hybrid nanostructures for advanced supercapacitors, Journal of Alloys and Compounds, Vol.775, pp. 1324-1356, 2019
- [8] Ali Saad, Zhixing Cheng, Hangjia Shen, Tiju Thomas, Minghui Yang, "Recent Advances in Nanocasting Cobalt-Based Mesoporous Materials for Energy Storage and Conversion", Electrocatalysi, Vol. 11, pp.465–484, 2020
- [9] <u>Hong Wang, Liyong Chen, Yuhua Feng, Hongyu Chen</u>, "Exploiting Core–Shell Synergy for Nanosynthesis and Mechanistic Investigation", *Acc. Chem. Res.* 2013, 46, 7, 1636– 1646, 2018.
- [10] Hao-peng Feng, Lin Tang, Guang-ming Zeng, Yaoyu Zhou, Yao-cheng Deng, Xiaoya Ren, Biao Song, Chao Liang, Mengyun Wei, Jiang-fang Yu, "Advances in Colloid and Interface Science", Vol. 267, Pp. 26-46, 2019
- [11] Aqib Muzaffar, M. Basheer Ahamed, Kalim Deshmukh, Jagannathan Thirumalai, "A review on recent advances in hybrid supercapacitors: Design, fabrication and applications", Renewable and Sustainable Energy Reviews, Vol. 101, pp. 123-145, 2019
- [12] Ahmed Afif, Sheikh MH Rahman, Atia Tasfiah Azad, Juliana Zaini, Md Aminul Islan, Abul Kalam Azad, "Advanced materials and technologies for hybrid supercapacitors for energy storage – A review", Journal of Energy Storage, Vol.25, pp. 100852, 2019
- [13] <u>B. E. Conway , W. G. Pell</u>, "Double-layer and pseudocapacitance types of electrochemical capacitors and their applications to the development of hybrid devices", Journal of Solid State Electrochemistry, Vol. 7, pp.637–644, 2003.
- [14] Jinfeng Li, Phi H. Q. Pham, Weiwei Zhou, Ted D. Pham, Peter J. Burke, 'Carbon-Nanotube–Electrolyte Interface: Quantum and Electric Double Layer Capacitance', ACS Nano, Vol. 12, Issue 10, pp. 9763–9774, 2018
- [15] Simon Fleischmann, James B. Mitchell, Ruocun Wang, Cheng Zhan, De-en Jiang, Volker Presser, Veronica Augustyn, 'Pseudocapacitance: From Fundamental Understanding to High Power Energy Storage Materials", *Chem. Rev.* Vol. 120, Issue 14, pp. 6738–6782, 2020
- [16] Dubal, D. P., O. Ayyad, V. Ruiz, and P. Gómez-Romero., "Hybrid energy storage: The merging of battery and supercapacitor chemistries". *Chemical Society Reviews* 44 (7):1777–90. doi:10.1039/C4CS00266K., 2015
- [17] <u>Muhammad Zahir Iqbal, Mian Muhammad Faisal</u>, "Integration of supercapacitors and batteries towards high-performance hybrid energy storage devices", Energy Research, Vol. 45, Issue 2, pp. 1449-1479,2020
- [18] <u>MaherF. El-Kady, Melanie Ihns, Mengping Li, Richard B. Kaner</u>, 'Engineering three-dimensional hybrid supercapacitors and microsupercapacitors for high-performance integrated energy storage", PNAS, Vol. 112, Issue 14), pp. 4233-4238, 2015
- [19] Yin, Y., Liu, C., Fan, S., "Hybrid energy storage devices combining carbon-nanotube/polyaniline supercapacitor with lead-

© 2024, IJSRMS All Rights Reserved

acid battery assembled through a "directly- inserted" method(Article)", RSC Adv., Vol. 4, pp. 26378-26382, 2014

- [20] P.A. Nelson, J.R. Owen , 'A High-Performance Supercapacitor/Battery Hybrid Incorporating Templated Mesoporous Electrodes(Article) ",J. Electrochem. Soc., Vol. 150, pp. A1313-A1317, 2003
- [21] Y.G. Wang, Y.Y. Xia ,' A new concept hybrid electrochemical surpercapacitor: Carbon/LiMn 2O₄ aqueous system(Article)", Electrochem. Commun., Vol. 7, pp. 1138-1142, 2005
- [22] Q. Wang, Z.H. Wen, J.H. Li, "A hybrid supercapacitor fabricated with a carbon nanotube cathode and a TiO₂-B nanowire anode(Article)", Adv. Funct. Mater., Vol. 16, pp. 2141-2146, 2006
- [23] G.C. Li, G.R. Li, S.H. Ye, X.P. Gao , "A Polyaniline-coated sulfur/carbon composite with an enhanced high-rate capability as a cathode material for lithium/sulfur batteries(Article)", Adv. Energy Mater., Vol. 2, pp. 1238-1245, 2012
- [24] Muhammad Sajjad ^{a b}, Muhammad Ibrar Khan ^c, Fang Cheng ^{a b}, Wen Lu, :A review on selection criteria of aqueous electrolytes performance evaluation for advanced asymmetric supercapacitors", Journal of Energy Storage, Vol. 40, pp. 102729, 2021.
- [25] Wenjun Zhou, Meng Zhang, Xiangyue Kong, Weiwei Huang, Qichun Zhang, "Recent Advance in Ionic-Liquid-Based Electrolytes for Rechargeable Metal-Ion Batteries", Advanced Science, Vol. 8, Issue 13, pp.2004490, 2021
- [26] Dan Gao, Zhiling Luo, Changhong Liu, Shoushan Fan, "A survey of hybrid energy devices based on supercapacitors", Green Energy & Environment, Vol. 8, Issue 4, pp. 972-988, 2023.
- [27] Y.L. Yin, C.H. Liu, S.S. Fan , "Hybrid energy storage devices combining carbon-nanotube/polyaniline supercapacitor with leadacid battery assembled through a "directly- inserted" method(Article)", RSC Adv., Vol. 4, pp. 26378-26382, 2004.
- [28] R. Bhattacharyya, B. Key, H.L. Chen, A.S. Best, A.F. Hollenkamp, C.P. Grey, "In situ NMR observation of the formation of metallic lithium microstructures in lithium batteries(Article)", Nat. Mater., Vol. 9, pp. 504-510, 2010.
- [29] P.A. Nelson, J.R. Owen , "A High-Performance Supercapacitor/Battery Hybrid Incorporating Templated Mesoporous Electrodes(Article)}, J. Electrochem. Soc., Vol. 150, pp. A1313-A1317, 2003.
- [30] Eunho Lim, Changshin Jo, Haegyeom Kim, Mok-Hwa Kim, Yeongdong Mun, Jinyoung Chun, Youngjin Ye, Jongkook Hwang, Kyoung-Su Ha, Kwang Chul Roh, Kisuk Kang, Songhun Yoon, Jinwoo Lee, "Facile Synthesis of Nb2O5@Carbon Core-Shell Nanocrystals with Controlled Crystalline Structure for High-Power Anodes in Hybrid Supercapacitors", ACS Nano, 9(7):7497-505. doi: 10.1021/acsnano.5b02601. Epub 2015 Jun 24, 2015.
- [31] <u>Kuo-Chuan Ho</u>, Lu-Yin Lin , "A review of electrode materials based on core-shell nanostructures for electrochemical supercapacitors," J. Mater. Chem. A, Vol. 7, pp. 3516-3530, 2019
- [32] Pascal Nbelayim, Yuya Ashida, Keiichiro Maegawa, Go Kawamura, Hiroyuki Muto, Atsunori Matsuda, "Preparation and Characterization of Stable and Active Pt@TiO₂ Core–Shell Nanoparticles as Electrocatalyst for Application in PEMFCs", ACS Appl. Energy Mater. Vol. 4, pp. 3269–3281, 2020.
- [33] Qi Liu, Zhe-Fei Li, Yadong Liu, Hangyu Zhang, Yang Ren, Cheng-Jun Sun, Wenquan Lu, Yun Zhou, Lia Stanciu, Eric A. Stach, Jian Xie, "Graphene-modified nanostructured vanadium pentoxide hybrids with extraordinary electrochemical performance for Li-ion batteries', <u>Nature Communications</u>, Vol. 6, pp.6127, 2015
- [34] Qingqing Ke, Minrui Zheng, Huajun Liu, Cao Guan, Lu Mao, John Wang, "3D TiO₂@Ni(OH)₂ Core-shell Arrays with Tunable Nanostructure for Hybrid Supercapacitor Application", <u>Scientific</u> <u>Reports</u> 5, 13940, 2015.
- [35] .Xibin Xu, Miao Liu, Jiang-shan Luo, Yu-ying Wang, "Nanoscale Energy Confinement and Hybridization of Surface Plasmons Based on Skin Depth in Au/Ag Core-Shell Nanostructures", <u>Plasmonics</u>, Vol. 10, Issue 4, pp. 797-808, 2015

- [36] Das, S.; Pérez-Ramírez, J.; Gong, J.; Dewangan, N.; Hidajat, K.; Gates, B.C.; Kawi, S. , 'Core–Shell Structured Catalysts for, Thermocatalytic, Photocatalytic, and Electrocatalytic Conversion of CO₂", Chem. Soc. Rev. ,Vol. 49, pp. 2937–3004, 2021.
- [37] Rimmy Singh, Rachna Bhateria, "Core-shell nanostructures: a simplest two-component", Environ Geochem Health, <u>https://doi.org/10.1007/s10653-020-00766-1(0</u>, 2020
- [38] 1<u>Kunfeng Chen, Shuyan Song</u>, Fei Liu, Dongfeng Xue, "Structural design of graphene for use in electrochemical energy storage devices", *Chem. Soc. Rev.*, Vol.44, pp.6230-6257, 2015.
- [39] Mayuresh Khot, Rahaman Sharif Shaik, Wania Touseef, <u>Amirkianoosh Kian</u>, "Binder-free NiO/CuO hybrid structure via ULPING (Ultra-short Laser Pulse for In-situ Nanostructure Generation) technique for supercapacitor electrode", <u>Scientific</u> Reports, Vol. 13, pp. 6975, 2023.
- [40] Yan Zhao, Linfeng Hu, Shuyan Zhao, Limin Wu, "Preparation of MnCo₂O₄@Ni(OH)₂ Core–Shell Flowers for Asymmetric Supercapacitor Materials with Ultrahigh Specific Capacitance", Advanced Functional Materials, <u>Vol. 26, Issue23</u>, pp. 4085-4093, 2016.
- [41] H. Vijeth ^{a b}, S.P. Ashokkumar ^a, L. Yesappa ^a, M. Vandana ^a, H. D evendrappa , "Hybrid core-shell nanostructure made of chitosan incorporated polypyrrole nanotubes decorated with NiO for allsolid-state symmetric supercapacitor application", Electrochimica Acta, Vol. 354, pp. 136651, 2020.
- [42] Liu Wan, Jiaxing Liu, Xiang Li, Yan Zhang, Jian Chen, Cheng Du, Mingjiang Xie , "Fabrication of core-shell NiMoO₄@MoS₂ nanorods for high-performance asymmetric hybrid supercapacitors", International Journal of Hydrogen Energy, Vol. 45, Issue 7, pp. 4521-4533, 2021.
- [43] Franciele Wolfart, a Deepak P. Dubal,*b Marcio Vidotti*a and Pedro G´omez-Romero, "Hybrid core-shell nanostructured electrodes made of polypyrrole nanotubes coated with Ni(OH)2 nanoflakes for high energy-density supercapacitors", RSC Adv., Vol. 6, pp. 15062, 2016.
- [44] Bell, Charleson Sherard, 'Novel multilayered magnetoplasmonic nanoparticles for theranostic applications", Vanderbilt University ProQuest Dissertations Publishing. pp.10582903, 2015
- [45] Qun Liu, Xiaodan Hong, Xingyan You, Xin Zhang, Xin Zhao, Xing Chen, Meidan Ye, Xiangyang Liu , "Designing heterostructured metal sulfide core-shell nanoneedle films as battery-type electrodes for hybrid supercapacitors, Energy Storage Materials, Vol.24,pp. 541-549, 2020.
- [46] Chuan Xia, Wei Chen, Xianbin Wang, Mohamed N. Hedhili, Nini Wei, , "Highly Stable Supercapacitors with Conducting Polymer Core-Shell Electrodes for Energy Storage Applications, Advanced Energy Materials, Vol.<u>5</u>, Issue <u>8</u>, pp. 1401805, 2015.
- [47] K.W. Wang, Ting Yan, W.G. Pan, "Optimization strategies of microencapsulated phase change materials for thermal energy storage", Journal of Energy Storage, Vol. 68, pp.107844, 2023.
- [48] .Rajib Ghosh Chaudhuri, Santanu Paria ," Core/Shell Nanoparticles: Classes, Properties, Synthesis Mechanisms, Characterization, and Applications", *Chem. Rev.* Vol. 4, pp. 2373– 2433, 2012.
- [49] Vancha Harish, M.M. Ansari, Devesh Tewari, Awadh Bihari Yadav, Neelesh Sharma, Sweta Bawarig María-Luisa García-Betancourt, Ali Karatutlu, Mikhael Bechelany, Ahmed Barhoum, "Cutting-edge advances in tailoring size, shape, and functionality of nanoparticles and nanostructures: A review", Journal of the Taiwan Institute of Chemical Engineers, Vol.149, pp.105010, 2023.
- [50] Karthik Kiran Sarigamala, Shobha Shukla, Alexander Struck, Sumit Saxena , "Rationally engineered 3D-dendritic cell-like morphologies of LDH nanostructures using graphene-based coreshell structures", Microsyst Nanoeng, Vol. 5, pp. 65, 2018
- [51] Yefeng Yang, Ding Cheng, Shaojie Chen, Yingli Guan, Jie Xiong ," Construction of Hierarchical NiCo₂S₄@Ni(OH)₂ Core-Shell Hybrid Nanosheet Arrays on Ni Foam for High-Performance Aqueous Hybrid Supercapacitors", Electrochimica Acta, Vol.193, pp.116-127, 2016.

- [52] Chunli Guo, Yan'an Meng, Deyang Yu, Liangyu Liu, Yingjie Hua, Xudong Zhao, Xiaoyang Liu, "Synthesis of the sandwich-type NiMn₂O₄@N-C@MnO₂ core-shell nanostructured materials for the high-performance battery-supercapacitor hybrid devices', Journal of Energy Storage, Vol. 68, pp. 107814,2023.
- [53] Xin Jiao, Biyu Li, Jian Wang, Yingchun Fan, Yongchang Ma, Zhihao Yuan, Chenguang Zhang ,' Size-controllable synthesis of covalently interconnected few-shelled Fe₃O₄@onion-like carbons for high-performance asymmetric supercapacitors, Carbon, Vol. 203, pp. 261-272, 2023.
- [54] Guihua Yu, Xing Xie, Lijia Pan, Zhenan Bao, Yi Cu, "Hybrid nanostructured materials for high-performance electrochemical capacitors", Nano Energy, Vol. 2, Issue 2,pp. 213-234, 2013.
- [55] Nagabandi Jayababu, Seungju Jo, Youngsu Kim, Daewon Kim "Preparation of NiO decorated CNT/ZnO core-shell hybrid nanocomposites with the aid of ultrasonication for enhancing the performance of hybrid supercapacitors", Ultrasonics Sonochemistry, Vol.71,pp. 105374, 2021.
- [56] Yefeng Yang, Ding Cheng, Shaojie Chen, Yingli Guan, Jie Xiong , "Construction of Hierarchical NiCo₂S₄@Ni(OH)₂ Core-Shell Hybrid Nanosheet Arrays on Ni Foam for High-Performance Aqueous Hybrid Supercapacitors", Electrochimica Acta, Vol.193, pp. 116-127, 2016
- [57] Songmin Zhang, Jiawen Wu, Jitong Wang, Wenming Qiao, Dongh ui Long, Licheng Ling ,:Constructing *T*-Nb₂O₅@Carbon hollow core-shell nanostructures for high-rate hybrid supercapacitor", Journal of Power Sources, Vol. 396, pp.88-94, 2018
- [58] Miao Du, Weimin Xia, Zhichao Jiao, Yuanqing Chen, Muslum Demir, Ying Zhang, Mengmeng Gu, Xiaoxuan Zhang, Cheng Wang, "Construction of hierarchical sugar gourd-like (Ni,Co)Se₂/(Ni,Co)Se₂/CC nanostructure with enhanced performance for hybrid supercapacitor, Journal of Alloys and Compounds, Vol. 930, pp.167459, 2023.
- [59] Huaping Zhao ," 3D Nanostructures for the Next Generation of High-Performance Nanodevices for Electrochemical Energy Conversion and Storage", Advanced Energy Materials, Vol.<u>10</u>, Issue <u>28</u>, pp. 2001460, 2020.
- [60] Arava Leela Mohana Reddy, Sanketh R. Gowda, Manikoth M. Shaijumon , "Hybrid Nanostructures for Energy Storage Applications, Advanced Materials, Vol. <u>24</u>, Issue <u>37</u>, pp. 5045-5064, 2012
- [61] Yulin Min , "Core-Shell and Yolk-Shell Nanocatalysts", Part of the Nanostructure Science and Technology book series (NST), pp 515–562, 2021.
- [62] <u>Xin Li, John Wang</u>, "One-dimensional and two-dimensional synergized nanostructures for high-performing energy storage and conversion", Infomat, <u>Vol. 2, Issue 1</u>, pp. 3-32, 2020
- [63] <u>Gen Chen, Litao Yan, Hongmei Luo</u>, Shaojun Guo, "Nanoscale Engineering of Heterostructured Anode Materials for Boosting Lithium-Ion Storage", Advanced Materials, <u>Vol. 28, Issue 35</u>, pp.7580-7602, 2016.
- [64] Long-bo Jiang, Xing-zhong Yuan, Jie Liang, Jin Zhang, Hou Wang Guang-ming Zeng ,"Nanostructured coreshell electrode materials for electrochemical capacitors", <u>Journal of</u> <u>Power Sources</u>, Volume 331, pp 408-425, 2016.
- [65] <u>Kuo-Chuan Ho</u>, Lu-Yin Lin, "A review of electrode materials based on core-shell nanostructures for electrochemical supercapacitors", J. Mater. Chem. A, Vol.7, pp.3516-3530, 2019
- [66] Ruiqi Liu¹, Shusheng Xu^{4,1}, Xiaoxuan Shao¹, Yi Wen¹, Xuerong Shi¹, Jing Hu² and Zhi Yang, "Carbon coating on metal oxide materials for electrochemical energy storage", *Nanotechnology* Vol.32 pp.502004, 2021.
- [67] Long-bo Jiang, Xing-zhong Yuan
 ^c, Jie Liang, Jin Zhang, Hou Wang, Guang-ming Zeng,
 "Nanostructured core-shell electrode materials for electrochemical capacitors", Journal of Power Sources, Vol.331, pp 408-425, 2016
- [68] Leiyun Han ^a, Xilong Liu ^a, Yang Chen ^a, Zheng Cui ^a, Yingjie Hua ^b, Chongtai Wang ^b, Xudong Zhao ^a, Xiaoyang Liu, "High energy density pouch-type supercapacitor achieved by MOFs derived 3D hollow N-doped carbon with Fe₂O₃ and hierarchical

CuCo₂S₄@NiFe-LDH core-shell nanostructures", Electrochimica Acta, Vol.467, pp. 143131, 2023

- [69] <u>Ronghao Wang, Hongmin Liu, Yuhao Zhang, Kaiwen Sun</u>, Weizhai Bao, "Integrated Photovoltaic Charging and Energy Storage Systems: Mechanism, Optimization, and Future", Nano-Micro Small, <u>Vol.18, Issue 31</u>, pp.2203014, 2022.
- [70] <u>Ximan Dong, Xinyue Chen, Xin Jiang</u>, Nianjun Yang, "Light-Assisted Energy Storage Devices: Principles, Performance, and Perspectives", Advanced Energy Materials, <u>Vol.13, Issue 38</u>, 2301143, 2023.
- [71] <u>Tao Liu, Liuyang Zhang, Bei Cheng</u>, Jiaguo Yu, "Hollow Carbon Spheres and Their Hybrid Nanomaterials in Electrochemical Energy Storage, Advanced Energy Materials, <u>Vol. 9, Issue 17</u>, pp.1803900, 2019.
- [72] Rui Xu, Lei Du, David Adekoya, Gaixia Zhang, Shanqing Zhang, Shuhui Sun , "Well-Defined Nanostructures for Electrochemical Energy Conversion and Storage", Advanced Energy Materials, Vol.11, Issue15, Special Issue: 10th Anniversary Edition, International Collaborations in Energy Research, pp.2001537, 2021.
- [73] Le Yu, <u>Han Hu, Hao Bin Wu</u>, "Complex Hollow Nanostructures: Synthesis and Energy-Related Applications", Advanced Materials, <u>Vol.29, Issue15</u>, pp.1604563, 2017.
- [74] Jing Pan, Shaobin Li, Fengbo Li, Wenzhi Zhang, Dongxuan Guo, Li Zhang, Deqing Zhang, Hong Pan, Yushu Zhang, Yifeng Ruan, "Design and construction of core-shell heterostructure of Ni-V layered double hydroxide composite electrode materials for high-performance hybrid supercapacitor and L-Tryptophan sensor", Journal of Alloys and Compounds, Vol. 890,pp.161781, 2022.
- [75] Hemming Zhao, Zepeng Zhang, Chungui Zhou, Huifang Zhang, "Turning the morphology and size of NiMoO₄ Nanosheets anchored on NiCo₂O₄ nanowires: the optimized core-shell hybrid for high energy density asymmetric supercapacitors", Applied surface science, Vol. 541, pp. 148458, 2021.
- [76] Danyang Li, Jing Lin, Yang Huang, Xin He, Chao Yu, Jun Zhang, Chengchun Tang, "MnO₂ Nanosheets grown on N-doped agaricderived three-dimensional porous carbon for asymmetric supercapacitors", Journal of alloys and compounds. Vol. 815, pp. 152344, 2020.
- [77] Qingqing Ke, Minrui Zhang, Huajun Liu, Cao Guan, Lu Mao, John Wang, "3D TiO₂@Ni(OH)₂ core-shell Arrays with Tunable Nanostructure for hybrid supercapacitor application", Scientific Reports, Vol. 5, pp. 13940, 2015.
- [78] Ashutosh Kumar Singh, Kalyan Mandal, 'Core-Shell hybrid Nanostructure Based High performance supercapacitor Electrode", 2 Physics, Indian, 2015.
- [79] Ashutosh k. Singh, Kalyan Mandal, "Engineering of high performance supercapacitor electrode based on Fe-Ni/Fe₂O₃-NiO core-shell Nanostructures", Journal of Applied Physics, Vol.117, pp. 105101, 2015.
- [80] Jianming Wang, Ying Huang, Xianping Du, Shuai Zhang, Meng Zang, "Hollow 1D carbon tube core anchored in Co₃O₄@SnS₂ multiple shells for constructing free electrodes, Chemical Engineering Journal, Vol.464, pp.142741, 2023.
- [81] Yi Wang, Jie Wang, Dong Wei, Lan Xu, "Multicore-shell MnO₂@ppy@N-doped porous carbon Nanofiber ternary composites as electrode materials for high-performance supercapacitors", Journal of Colloid and interface Science, Vol. 648, pp. 925-939, 2023.
- [82] Yumak, T., Bragg, D., &Sabolsky, E. M. "Effect of synthesis methods on the surface and electrochemical characteristics of metal oxide/activated carbon composites for supercapacitor applications". *Applied Surface Science*, Vol. 469, pp. 983-993, 2019.
- [83] Chang, C., Chen, W., Chen, Y., Chen, Y., Ding, F, Liu, Z, "Recent progress on two-dimensional materials". Acta Phys.-Chim. Sin, Vol.37, Issue 12, pp.2108017, 2021.
- [84] Zardkhoshoui, A. M, Davarani, S. S. H. "Designing a flexible allsolid-state supercapacitor based on CuGa2O4 and FeP-rGO

electrodes". Journal of Alloys and Compounds, Vol. 773, pp.527-536, 2019.

- [85] Changhoon Song Junyeong Yun, Kayeon Keum, Yu Ra Jeong, Heun Park, Hanchan Lee, Geumbee Lee, Seung Yun Oh, Jeong Sook Ha, "High performance wire-type supercapacitor with Ppy/CNT-ionic liquid/AuNP/carbon fiber electrode and ionic liquid based electrolyte", Carbon, Vol. 144,pp. 639-648, 2019.
- [86] Liu, J., Qiao, S. Z., Budi Hartono, S., & Lu, G. Q. "Monodisperse yolk-shell nanoparticles with a hierarchical porous structure for delivery vehicles and nanoreactors". Angewandte Chemie International Edition, Vol.49, Issue 29, pp. 4981–4985, 2010.
- [87] Kamata, K., Lu, Y., & Xia, Y, "Synthesis and characterization of monodispersed core- shell spherical colloids with movable cores". Journal of the American Chemical, Society, Vol. 125, Issue 9, pp. 2384–2385, 2003.
- [88] Kim, J. Y., Yoon, S. B., Kooli, F., Lee, C. W., &Yu, J. S, "Synthetic control of ordered and disordered arrays of carbon nanofibers from SBA-15 silica templates". Chemical Communications, Vol. 14, pp. 1740–1741, 2003.
- [89] Yin, Y., Rioux, R. M., Erdonmez, C. K., Hughes, S., Somorjai, G. A., & Alivisatos, A. P, "Formation of hollow nanocrystals through the nanoscale Kirkendall effect", Science, Vol. 304, Issue 5671, pp. 711–714, 2004.
- [90] Camargo, P. H., Xiong, Y., Ji, L., Zuo, J. M., & Xia, Y, "Facile synthesis of tadpole-like nanostructures consisting of Au heads and Pd tails". Journal of the American Chemical Society, Vol. 129, Issue 50, pp. 15452–15453, 2007.
- [91] Zhang, T., Ge, J., Hu, Y., Zhang, Q., Aloni, S.,&Yin, Y., "Formation of hollow silica colloids through a spontaneous dissolution-regrowth process. Angewandte Chemie International Edition, Vol. 47, Issue 31, pp. 5806–5811, 2008.
- [92] Liu, J., Xu, J., Che, R., Chen, H., Liu, M., & Liu, Z. "Hierarchical Fe3O4@ TiO2 yolk-shell microspheres with enhanced microwaveabsorption properties". Chemistry A European Journal, Vol. 19, Issue 21, pp. 6746–52, 2013.
- [93] Liu, R., Guo, Y., Odusote, G., Qu, F., & Priestley, R. D. "Coreshell Fe3O4 polydopamine nanoparticles serve multipurpose as drug carrier, catalyst support and carbon adsorbent". ACS applied materials & interfaces, Vol. 5, Issue 18, pp. 9167–9171, 2013.
- [94] Zhang, L., Wang, T., Li, L., Wang, C., Su, Z., & Li, J. "Multifunctional fluorescent-magnetic polyethyleneimine functionalized Fe3O4–mesoporous silica yolk–shell nanocapsules for siRNA delivery". Chemical Communications, Vol. 48, Issue 69, pp. 8706–8708, 2012.
- [95] Zhang, H., Zhou, L., Noonan, O., Martin, D. J., Whittaker, A. K., & Yu, C, "Tailoring the void size of iron oxide@ carbon yolk-shell structure for optimized lithium storage"., Advanced Functional Materials, Vol. 24, Issue 27, pp. 4337–4342, 2014.
- [96] Zhang, L., Wang, T., Li, L., Wang, C., Su, Z., & Li, J., "Multifunctional fluorescent-magnetic polyethyleneimine functionalized Fe3O4–mesoporous silica yolk–shell nanocapsules for siRNA delivery". Chemical Communications, Vol. 48, Issue 69, pp. 8706–8708, 2012.
- [97] Wu, X. J., & Xu, D., "Formation of yolk/SiO2 shell structures using surfactant mixtures as template". Journal of the American Chemical Society, Vol. 131, Issue 8, pp. 2774–2775, 2009.
- [98] Velikov, K. P., & van Blaaderen, A., "Synthesis and characterization of monodisperse core- shell colloidal spheres of zinc sulfide and silica". Langmuir, Vol. 17, Issue 16, pp. 4779– 4786, 2001.
- [99] Park, J., Joo, J., Kwon, S. G., Jang, Y., & Hyeon, T. "Synthesis of monodisperse spherical nanocrystals", Angewandte Chemie International Edition, Vol. 46, Issue 25, pp.4630–4660, 2007.
- [100] Zhang, J., Tang, Y., Weng, L., & Ouyang, M., "Versatile strategy for precisely tailored core@ shell nanostructures with single shell layer accuracy: The case of metallic shell", Nano letters, Vol. 9, Issue 12, pp. 4061–4065, 2009.
- [101] Muzikansky, A, Nanikashvili, P., Grinblat, J., & Zitoun, D., "Ag dewetting in Cu@ Ag monodisperse core-shell nanoparticles". The Journal of Physical Chemistry C, Vol. 117, Issue 6, pp. 3093–3100, 2013.

- [102] Chen, D., Xia, X., Gu, H., Xu, Q., Ge, J., Li, Y., & Lu, J. "PHresponsive polymeric carrier encapsulated magnetic nanoparticles for cancer targeted imaging and delivery", Journal of Materials Chemistry, Vol. 21, Issue 34, pp.12682–12690, 2011.
- [103] Chen, L., Xu, S., & Li, J. "Recent advances in molecular imprinting technology: Current status, challenges and highlighted applications". Chemical Society Reviews, Vol. 40, Issue 5, pp.2922–2942, 2011.
- [104] Shiomi, T., Matsui, M., Mizukami, F., & Sakaguchi, K. 'A method for the molecular imprinting of hemoglobin on silica surfaces using silanes". Biomaterials, Vol. 26, Issue27, pp.5564– 5571, 2005.
- [105] Jia, X., Li, H., Luo, J., Lu, Q., Peng, Y., Shi, L., et al. "Rational design of core-shell molecularly imprinted polymer based on computational simulation and Doehlert experimental optimization: Application to the separation of tanshinone IIA from Salvia miltiorrhiza Bunge". Analytical and Bioanalytical Chemistry, Vol. 403, Issue 9, pp. 2691–2703, 2012.
- [106] Jankiewicz, B. J., Jamiola, D., Choma, J.,&Jaroniec, M. "Silicametal core-shell nanostructures". Advances in colloid and interface science, Vol. 170, Issue 1–2, pp. 28–47, 2012.
- [107] Gustafsson, G., Cao, Y., Treacy, G. M., Klavetter, F., Colaneri, N., & Heeger, A. J, 'Flexible light-emitting diodes made from soluble conducting polymers". Nature, Vol.357, Issue 6378, pp.477, 1992.
- [108] Liu, W., Kumar, J., Tripathy, S., Senecal, K. J., & Samuelson, L. "Enzymatically synthesized conducting polyaniline". Journal of the American Chemical Society, Vol. 121, Issue 1, pp. 71–78, 1999.
- [109] Holleman, A. F., & Wiberg, E. "Inorganic chemistry. Berlin/New York: Academic Press, 2001.
- [110] Liz-Marzan, L. M., Correa-Duarte, M. A., Pastoriza-Santos, I., Mulvaney, P., Ung, T., Giersig, M., et al. "Coreshell nanoparticles and assemblies thereof. In H. S. Nalawa (Ed.), Hand book of surfaces and interfaces of materials". Amsterdam: Elsevier, 2001
- [111] Ung, T., Liz-Marza'n, L. M., & Mulvaney, P. ,"Controlled method for silica coating of silver colloids. Influence of coating on the rate of chemical reactions", Langmuir, Vol.14, Issue 14, pp. 3740–3748, 1998.
- [112] Xie, L., Guo, J., Zhang, Y., & Shi, S, "Efficient determination of protocatechuic acid in fruit juices by selective and rapid magnetic molecular imprinted solid phase extraction coupled with HPLC", Journal of Agriculture and Food Chemistry, Vol. 62, Issue 32, pp. 8221–8228, 2014.
- [113] Lu, F., Li, H., Sun, M., Fan, L., Qiu, H., Li, X., &Luo, C, "Flow injection chemiluminescence sensor based on coreshell magnetic molecularly imprinted NPs for determination of sulfadiazine". Analytica Chimica Acta, Vol.718, pp. 84–91, 2012.
- [114] Uzuriaga-Sanchez, R. J., Khan, S., Wong, A., Picasso, G., Pividori, M. I., & Sotomayor, M. D. T., "Magnetically separable polymer (magMIP) for selective analysis of biotin in food samples", Food Chemistry, Vol. 190, pp. 460–467, 2016.
- [115] Kan, X., Zhao, Q., Shao, D., Geng, Z., Wang, Z., & Zhu, J. J. ," Preparation and recognition properties of bovine hemoglobin magnetic molecularly imprinted polymers", Journal of Physical Chemistry B, Vol.114, Issue 11, pp. 3999–4004, 2010.
- [116] Pichon, B. P., Gerber, O., Lefevre, C., Florea, I., Fleutot, S., Baaziz, W., & Pierron-Bohnes, V., "Microstructural and magnetic investigations of wustite-spinel core-shell cubic-shaped nanoparticles". Chemistry of Materials, Vol.23, Issue 11, pp. 2886– 2900, 2011.
- [117] Gao, Q., Chen, F., Zhang, J., Hong, G., Ni, J., Wei, X., & Wang. D, "The study of novel Fe3O4@ c-Fe2O3 core/shell nanomaterials with improved properties", Journal of magnetism and magnetic materials, Vol. 321, Issue 8, pp. 1052–1057, 2009.
- [118] Tian, Y., Wu, D., Jia, X., Yu, B., Zhan, S, "Core-Shell Nanostructure of Synthesis and Photocatalysis for Methyl Orange". Journal of Nanomaterials, 2011
- [119] Wang, H., Chen, L., Feng, Y., & Chen, H., "Exploiting core-shell synergy for nanosynthesis and mechanistic investigation". Accounts of chemical research, Vol. 46, Issue 7, pp. 1636–1646, 2013.

- [120] Watson, K. J., Zhu, J., Nguyen, S. T., & Mirkin, C. A.,"Hybrid nanoparticles with block copolymer shell structures". Journal of the American Chemical Society, Vol. 121, Issue 2, pp. 462–463, 1999.
- [121] Willner, I., & Katz, E, "Magnetic control of electrocatalytic and bioelectrocatalytic processes", Angewandte Chemie, Vol.115, Issue 38, pp.4724–4737, 2003.
- [122] Xu, X., Friedman, G., Humfeld, K. D., Majetich, S. A., & Asher, S. A., "Superparamagnetic photonic crystals". Advanced materials, 1Vol. 3, Issue 22, pp. 1681–1684, 2001.
- [123] Xu, X., Friedman, G., Humfeld, K. D., Majetich, S. A., & Asher, S. A, "Synthesis and utilization of monodisperse superparamagnetic colloidal particles for magnetically controllable photonic crystals". Chemistry of Materials, Vol. 14, Issue 3, pp. 1249–1256, 2002.
- [124] Chen, D., Xia, X., Gu, H., Xu, Q., Ge, J., Li, Y., & Lu, J, "PHresponsive polymeric carrier encapsulated magnetic nanoparticles for cancer targeted imaging and delivery". Journal of Materials Chemistry, Vol. 21, Issue 34, pp. 12682–12690, 2011.
- [125] Chen, L., Xu, S., & Li, J, "Recent advances in molecular imprinting technology: Current status, challenges and highlighted applications". Chemical Society Reviews, Vol. 40, Issue 5, pp. 2922–2942, 2011.
- [126] Kim, H., Achermann, M., Balet, L. P., Hollingsworth, J. A., & Klimov, V. I, "Synthesis and characterization of Co/CdSe core/shell nanocomposites: Bifunctional magneticoptical nanocrystals", Journal of the American Chemical Society, Vol.127, 2, pp.544–546, 2005.
- [127] Kim, J. Y., Yoon, S. B., Kooli, F., Lee, C. W., &Yu, J. S. (2003). Synthetic control of ordered and disordered arrays of carbon nanofibers from SBA-15 silica templates. Chemical Communications, Vol. 14, pp.1740–1741.
- [128] Rocha, N., Mendes, J., Dura^es, L., Maleki, H., Portugal, A., Geraldes, C. F., & Coelho, J, "Poly (ethylene glycol)-block-poly (4vinyl pyridine) as a versatile block copolymer to prepare nanoaggregates of superparamagnetic iron oxide nanoparticles", Journal of Materials Chemistry B, Vol.2, Issuel1, pp.1565–1575, 2014.
- [129] Serpell, C. J., Cookson, J., Ozkaya, D., & Beer, P. D, "Core@ shell bimetallic nanoparticle synthesis via anion coordination", Nature chemistry, Vol. 3, Issue 6, pp. 478, 2011.
- [130] Mizukoshi, Y., Fujimoto, T., Nagata, Y., Oshima, R., & Maeda, Y, "Characterization and catalytic activity of coreshell structured gold/palladium bimetallic nanoparticles synthesized by the sonochemical method", The Journal of Physical Chemistry B, Vol. 104, Issue 25, pp. 6028–6032, 2000.
- [131] Yan, J. M., Zhang, X. B., Akita, T., Haruta, M.,&Xu, Q, "Onestep seeding growth of magnetically recyclable Au@Co core- shell nanoparticles: Highly efficient catalyst for hydrolytic dehydrogenation of ammonia borane", Journal of the American Chemical Society, Vol. 132, Issue 15, pp. 5326–5327, 2010.
- [132] Zheng, Y., Cheng, Y., Wang, Y., Bao, F., Zhou, L., Wei, X., et al., "Quasicubic a-Fe2O3 nanoparticles with excellent catalytic performance", The Journal of Physical Chemistry B, Vol. 110, Issue 7, pp. 3093–3097, 2006.
- [133] V. Schmidt, P. C. McIntyre, and U. Gösele, "Morphological instability of misfit-strained core-shell nanowires", Phys. Rev. B Vol. 77, pp. 235302, 2008.
- [134] <u>D. Bochicchio, R. Ferrando</u>, "Morphological instability of coreshell metallic nanoparticles, Materials Science, Physics, DOI:<u>10.1103/PhysRevB.87.165435</u>, 2013.
- [135] <u>Shai Mangel, L. Houben, M. Bar Sadan</u>, "The effect of atomic disorder at the core-shell interface on stacking fault formation in hybrid nanoparticles", Nanoscale, Vol. 8, pp.17568, 2016
- [136] <u>Hong Wang, Liyong Chen, Yuhua Feng, Hongyu Chen</u>, "Exploiting core-shell synergy for nanosynthesis and mechanistic investigation", Acc Chem Res, Vol. 16;46, Issue 7, pp. 1636-46, 2013
- [137] Yassmin Ibrahim, Ahmed Mohamed, Ahmed M. Abdelgawad, Kamel Eid, Aboubakr M. Abdullah, Ahmed Elzatahry, "The Recent Advances in the Mechanical Properties of Self-Standing Two-Dimensional MXene-Based Nanostructures: Deep Insights

into the Supercapacitor", Nanomaterials, Vol. 10, Issue 10, pp. 1916, 2020

- [138] Zhengchen Wu, Han-Wen Cheng, Chen Jin, Bintong Yang, Chunyang Xu, Ke Pei, Huibin Zhang, Ziqi Yang, Renchao Che, "Dimensional Design and Core–Shell Engineering of Nanomaterials for Electromagnetic Wave Absorption", Advanced, Advanced Materials, <u>Vol. 34, Issue11</u>, pp. 2107538, 2021.
- [139] Xin LXini John Wang, "One-dimensional and two-dimensional synergized nanostructures for high-performing energy storage and conversion", InfoMat, <u>Vol. 2, Issue1</u>, pp. 3-32, 2019
- [140] Qi Xue, Jinfeng Sun, Yan Huang, Minshen Zhu, Zengxia Pei, Hongfei Li, Yukun Wang, Na Li, Haiyan Zhang, Chunyi Zhi, "Recent Progress on Flexible and Wearable Supercapacitors", Nano-Micro Small, Vol. 13, Issue 45, pp. 1701827, 2017
- [141] Liyang Lin, Huiming Ning, Shufeng Song, Chaohe Xu, Ning H u, "Flexible electrochemical energy storage: The role of composite materials", Composites Science and Technology, Vol.192, pp.108102, 2020.
- [142] Guangmeng Qu^a, Pengxiao Sun^a, Guotao Xiang^a, Jiangmei Yin^a, Qin Wei^{b c}, Chenggang Wang^a, Xijin Xu, "Moss-like nickel-cobalt phosphide nanostructures for highly flexible all-solid-state hybrid supercapacitors with excellent electrochemical performances, Vol. 20, pp. 100713, 2020
- [143] Ganest Kumar Veerasubramani, <u>Arunkumar</u> <u>Chandrasekhar, Sudhakaran M. S. P., Young Sun Mok, Sang Jae</u> <u>Kim</u>, "Liquid electrolyte mediated flexible pouch-type hybrid supercapacitor based on binderless core-shell nanostructures assembled with honeycomb-like porous carbon", J. Mater. Chem. A, Vol. 5, pp. 11100-11113, 2017
- [144] Humin Zheng , <u>Teng Zhai</u>,^a <u>Minghao Yu</u>,^a <u>Shilei</u> Xie,^a <u>Chaolun Liang</u>,^{ad} <u>Wenxia Zhao</u>,^d <u>Shing Chi Ian</u> <u>Wang</u>,^e <u>Zishou Zhang</u>*^{ac} and <u>Xihong Lu</u>, "TiO₂@C core–shell nanowires for high-performance and flexible solid-state supercapacitors", J. Mater. Chem. C, Vol. 1, pp. 225-229, 2013.
- [145] Mianomiao Liang Mingshu Zhao, Haiyang Wang, Qingyang Zheng, Xiaoping Song ,"Superior cycling stability of a crystalline/amorphous Co3S4 core-shell heterostructure for aqueous hybrid supercapacitors", J. Mater. Chem. A, Vol. 6, pp. 21350-21359, 2018.
- [146] Mengqiao Wang, Zhaoqiang Li, Chengxiang Wang, Ruizheng Zhao, Caixia Li, Dexiang Guo, Luyuan Zhang, Longwei Yin, " Advanced Functional Materials, Novel Core–Shell FeOF/Ni(OH)₂ Hierarchical Nanostructure for All-Solid-State Flexible Supercapacitors with Enhanced Performance, <u>Vol. 27</u>, <u>Issue 31</u>, pp. 1701014, 2017.
- [147] Hao-peng Feng, Lin Tang , Guang-ming Zeng , Yaoyu Zhou , Yao-cheng Deng , Xiaoya Ren , Biao Song , Chao Liang, Meng-yun Wei , Jiang-fang Yu , "Core-shell nanomaterials: Applications in energy storage and conversion", Advances in Colloid and Interface Science, Vol. 267, pp. 26-46, 2019.
- [148] Qiwen Ran, Hongyuan Zhao, Youzuo Hu, Shuai Hao, Jintao Li u, Hao Li, Xingquan Liu, "Enhancing surface stability of LiNi_{0.8}Co_{0.1}Mn_{0.1}O₂ cathode with hybrid core-shell nanostructure induced by high-valent titanium ions for Li-ion batteries at high cut-off voltage", Journal of Alloys and Compounds, Vol. 834, pp 155099, 2020.
- [149] Wenhao Ren, Chenfeng Ding, Xuewei Fu, Yun Huang, "Advanced gel polymer electrolytes for safe and durable lithium metal batteries: Challenges, strategies, and perspectives", Energy Storage Materials, Vol. 34, pp. 515-535, 2021.
- [150] Hao-Peng Feng, <u>Lin Tang, Guang-ming Zeng, Jing Tang, Yao-cheng Deng, Ming Yan, Ya-ni Liu, Yao-yu Zhou, Xiao-ya Ren, Song Chen, "Carbon-based core-shell nanostructured materials for electrochemical energy storage", J. Mater. Chem. A, Vol. 6, pp. 7310-7337, 2018.</u>
- [151] Haitao Bi, Gang Sui, Xiaoping Yang, "Studies on polymer nanofibre membranes with optimized core-shell structure as outstanding performance skeleton materials in gel polymer electrolytes", Journal of Power Sources, Vol. 267,pp. 309-315, 2014.

© 2024, IJSRMS All Rights Reserved

- [152] Xiaolin Li¹, Wen Liu, Minye Zhang², Yiren Zhong, Zhe Weng, Yingying Mi², Yu Zhou, Min Li, Judy J Cha, Zhiyong Tang³, Hong Jiang², Xueming Li¹, Hailiang Wang, "Strong Metal-Phosphide Interactions in Core-Shell Geometry for Enhanced Electrocatalysis", Nano Lett, Vol. 17, Issue 3, pp. 2057-2063, 2017.
- [153] Chunjoong Kim, Patrick J. Phillips, Linping Xu, Angang Dong, Raffaella Buonsanti, Robert F. Klie, <u>Jordi Cabana</u>, "Stabilization of Battery Electrode/Electrolyte Interfaces Employing Nanocrystals with Passivating Epitaxial Shells, *Chem. Mater.* Vol. 27, Issue 1,pp. 394–399, 2015.
- [154] <u>Hao Shan, Wenpei Gao, Yalin Xiong, Fenglei Shi, Yucong Yan, Yanling Ma, Wen Shang, Peng Tao, Chengyi Song, Tao Deng, Hui Zhang, Deren Yang, Xiaoqing Pan, Jianbo Wu, "Nanoscale kinetics of asymmetrical corrosion in core-shell nanoparticles", *Nature Communications*, vol. 9, pp.1011, 2018.</u>
- [155] Qiao Zhang, Ilkeun Lee, Ji Bong Joo, Francisco Zaera, Yadong Yin, "Core-shell nanostructured catalysts", Acc Chem Res, Vol. 46, Issue 8, pp. 1816-24, 2013
 [156] Robert V Dennis^{5,1,2}, Vikas Patil^{5,3}, Justin L Andrews^{1,2}, Jeffrey P
- [156] Robert V Dennis^{5,1,2}, Vikas Patil^{5,3}, Justin L Andrews^{1,2}, Jeffrey P Aldinger⁴, Ganapati D Yadav³ and Sarbajit Banerjee, "Hybrid nanostructured coatings for corrosion protection of base metals: a sustainability perspective", <u>Materials Research Express</u>, Vol. 2, pp. <u>3</u>, 2015
- [157] <u>Rimmy Singh</u>, <u>Rachna Bhateria</u>, "Core-shell nanostructures: a simplest two-component system with enhanced properties and multiple applications", Environmental Geochemistry and Health, Vol.43, pp.2459–2482, 2021
- [158] Linhua Hu, Philipp Brüner, Thomas Grehl, Hidde H. Brongersma, Jordi Cabana, "Control of Chemical Structure in Core–Shell Nanocrystals for the Stabilization of Battery Electrode/Electrolyte Interfaces", *Chem. Mater.* Vol. 29, Issue 14, pp. 5896–5905, 2017.
- [159] <u>Roberto C Longo¹, Chaoping Liang¹, Fantai Kong¹</u>, Kyeongjae Cho, "Core-Shell Nanocomposites for Improving the Structural Stability of Li-Rich Layered Oxide Cathode Materials for Li-Ion Batteries, ACS Appl Mater Interfaces, Vol. 10, Issue 2, pp. 19226-19234, 2018
- [160] Daniel Göhl, Aaron Garg, Paul Paciok, Karl J. J. Mayrhofer, Marc Heggen, Yang Shao-Horn, Rafal E. Dunin-Borkowski, Yuriy Román-Leshkov, <u>Marc Ledendecker</u>, "Engineering stable electrocatalysts by synergistic stabilization between carbide cores and Pt shells", <u>Nature Materials</u> vol. 19, pp.287–291, 2020
- [161] Xianbin Liu, Ziping Wu, Yanhong Yin, "Hierarchical NiCo₂S₄@PANI core/shell nanowires grown on carbon fiber with enhanced electrochemical performance for hybrid supercapacitors", Chemical Engineering Journal, Vol. 323, pp. 330-339, 2017
- [162] Jinghuang Lin^a, Henan Jia^a, Haoyan Ling^a, Shulin Chen^a, Yifei Cai^a, Junlei Qi^a, Chaoqun Qu^b, Jian Cao^a, Weidong Fei^a, Jicai F eng, "Hierarchical CuCo₂S₄@NiMn-layered double hydroxide core-shell hybrid arrays as electrodes for supercapacitors", Chemical Engineering Journal, Vol. 336, pp. 562-569, 2018.
- [163] Yang Han^a, Qin Zhang^a, Nantao Hu^a, Xue Zhang^a, Yiyong M ai^a, Jiaqiang Liu^b, Xiaolin Hua^b, Hao Wei, "Core-shell nanostructure of single-wall carbon nanotubes and covalent organic frameworks for supercapacitors", Chinese Chemical Letters, Vol. 28, Issue 12, pp. 2269-2273, 2017
- [164] Ting Zhu, Jing Wang, Ghim Wei Ho, "Self-supported yolk-shell nanocolloids towards high capacitance and excellent cycling performance", Nano Energy, Vol. 18, pp.273-282, 2015.
- [165] <u>Grégory Spataro¹</u>, <u>Yohan Champouret¹</u>, <u>Yannick Coppel¹</u>, Myrtil L Kahn, "Prominence of the Instability of a Stabilizing Agent in the Changes in Physical State of a Hybrid Nanomaterial", Chemphyschem, Vol. 21, Issue 21, pp. 2454-2459, 2020.
- [166] D. Bowles, James Y. Shen, "Thermal cycling effects on the dimensional stability of P75 and P75-T300 (fabric) hybrid graphite/epoxy laminates", Materials Science, Engineering, 1983.
- [167] Zakaria Razak, Abu Bakar Sulong, Norhamidi Muhamad, Che Hassan Che Haron, Mohd Khairul Fadzly Md Radzi, Nur Farhani Ismail, Dulina Tholibon, Izdihar Tharazi, "Effects of thermal cycling on physical and tensile properties of injection moulded kenaf/carbon nanotubes/polypropylene hybrid

composites", Composites Part B: Engineering, Vol.168, pp.159-165, 2019.

- [168] Yan Yao, <u>Nian Liu</u>, <u>Matthew T. McDowell</u>, Mauro Pasta and <u>Yi Cui</u>, "Improving the cycling stability of silicon nanowire anodes with conducting polymer coatings", Energy Environ. Sci., Vol. 5, pp. 7927-7930, 2012.
- [169] <u>Hamed Akbarzadeh¹</u>, Esmat Mehrjouei¹, Amir Nasser Shamkhali, "Au@Void@Ag Yolk-Shell Nanoclusters Visited by Molecular Dynamics Simulation: The Effects of Structural Factors on Thermodynamic Stability", J Phys Chem Lett, Vol. 8, Issue 13, pp. 2990-2998, 2017
- [170] Martin Schnedlitz¹, Ricardo Fernandez-Perea², Daniel Knez³, Maximilian Lasserus¹, Alexander Schiffmann¹, Ferdinand Hofer³, Andreas W Hauser¹, Maria Pilar de Lara-Castells⁴, Wolfgang E Ernst, "Effects of the Core Location on the Structural Stability of Ni-Au Core-Shell Nanoparticles", J Phys Chem C Nanomater Interfaces, Vol. 123, Issue 32, pp.20037-20043, 2019.
- [171] <u>Max Willinger¹</u>, Erik Reimhult, "Thermoresponsive Nanoparticles with Cyclic-Polymer-Grafted Shells Are More Stable than with Linear-Polymer-Grafted Shells: Effect of Polymer Topology, Molecular Weight, and Core Size', J Phys Chem B, Vol.125, Issue 25, pp. 7009-7023, 2021
- Vol.125, Issue 25, pp. 7009-7023, 2021
 [172] Hongxiang Wang¹, Hucheng Song¹, Zixia Lin¹, Xiaofan Jiang¹, Xiaowei Zhang¹, Linwei Yu¹, Jun Xu¹, Lijia Pan¹, Junzhuan Wang¹, Mingbo Zheng¹, Yi Shi¹, Kunji Chen, "Highly cross-linked Cu/a-Si core-shell nanowires for ultra-long cycle life and high rate lithium batteries", Nanoscale, Vol.8, Issue 5, pp.2613-9, 2016
- [173] Hao-peng Feng , Lin Tang , Guangming Zeng , Yaoyu Zhou , Yao cheng Deng , Xiaoya Ren , Biao Song , Chao Liang , Meng-yun Wei , Jiang-fang Yu , "Coreshell nanomaterials: Applications in energy storage and conversion", Advances in Colloid and Interface Science, Vol. 267, pp. 26-46, 2019.
- [174] <u>Kuo-Chuan Ho</u>, Liu-Yin Lin," A review of electrode materials based on core-shell nanostructures for electrochemical supercapacitors", J. Mater. Chem. A, Vol. 7, pp. 3516-3530, 2019.
- [175] Li Liu, Yuehong Qi, Jinrong Lu, Shuanglong Lin, Weijia An, Yin ghua Liang, Wenquan Cui, "A stable Ag₃PO₄@g-C₃N₄ hybrid core@shell composite with enhanced visible light photocatalytic degradation", Applied Catalysis B: Environmental, Vol. 183, pp. 133-141, 2016
- [176] Wei Li, Ahmed Elzatahry, Dhaifallah Aldhayan, Dongyuan Zhao, "Core-shell structured titanium dioxide nanomaterials for solar energy utilization", Chem. Soc. Rev., Vol. 47, pp.8203-8237, 2018
- [177] Tilahun Temesgen, Eneyew Tilahun Bekele, Bedasa Abdisa Gonfa, Lemma Teshome Tufa, Fedlu Kedir Sabir, Sisay Tadesse, Yilkal Dessie, "Advancements in biomass derived porous carbon materials and their surface influence effect on electrode electrochemical performance for sustainable supercapacitors: A review", Journal of Energy Storage, Vol.73, Part D, pp. 109293, 2023.
- [178] Yuerong Zhou^a, Yunhe Li^a, Yilong Hou^a, Ce Wang^a, Ying Yang ^a, Jiangwei Shang^b, Xiuwen Cheng, "Core-shell catalysts for the elimination of organic contaminants in aqueous solution: A review", Chemical Engineering Journal, Vol.455, pp.140604, 2023.
- [179] <u>Kai Qi, Ruizuo Hou, Shahid Zaman, Bao Yu Xia</u>, Hongwei, "A core/shell structured tubular graphene nanoflake-coated polypyrrole hybrid for all-solid-state flexible supercapacitors", J. Mater. Chem. A, Vol. 6,pp. 3913-3918, 2018
- [180] T. Kavinkumar, Selvaraj Seenivasan, Hong H. Lee, Hyeonjung Jung, Jeong Woo Han, Do-Heyoung Kim, "Interface-modulated uniform outer nanolayer: A category of electrodes of nanolayer-encapsulated core-shell configuration for supercapacitors", Nano Energy, Vol.81, pp.105667, 2021
- [181] Lingli Cheng, Yiyang Hu, Lei Ling, Dandan Qiao, Shicong Cui, Zheng Jiao," One-step controlled synthesis of hierarchical hollow Ni₃S₂/NiS@Ni₃S₄ core/shell submicrospheres for high-performance supercapacitors", Electrochimica Acta, Vol. 283, pp. 664-675, 2018.

© 2024, IJSRMS All Rights Reserved

[182] Long-bo Jiang , Xing-zhong Yuan

, Jie Liang, Jin Zhang , Hou Wang , Guang-ming Zeng , "Nanostructured core-shell electrode materials for electrochemical capacitors", Journal of Power Sources, Vol. 331, pp. 408-425, 2016.

- [183] Fengqi Liu, Chenbo He, Yonggang Jiang, Yaping Yang, Fei Pe ng ^a, Lanfang Liu ^a, Jing Men ^a, Junzong Feng, Liangjun Li, Guihu a Tang, Jian Feng, "Carbon layer encapsulation strategy for designing multifunctional core-shell nanorod aerogels as hightemperature thermal superinsulators", Chemical Engineering Journal, Vol. 455, pp.140502, 2021.
- [184] Yu Zhang, Benedict You Wei Hsu, Changliang Ren, Xu Li and John Wang (2015), "Silica-based nanocapsules: synthesis, structure control and biomedical applications", Chem. Soc. Rev, vol.44, pp.315-335
- [185] Duy Thanh Tran, Tolendra Kshetri, Dinh Chuong Nguyen, Jagadis Gautam, Van Hien Hoa, Huu Tuan Le, Nam Hoon Kim, Joong Hee Lee, "Emerging core-shell nanostructured catalysts of transition metal encapsulated by twodimensional carbon materials for electrochemical applications", Nanotoday, Vol. 22, pp. 100-131, 2018
- [186] Wanlu Yang, <u>Zan Gao</u>, Jing Ma,^a <u>Xingming Zhang Jun</u> <u>Wang</u> and Jingyuan Liu, "Hierarchical NiCo₂O₄@NiO core-shell hetero-structured nanowire arrays on carbon cloth for a high-performance flexible all-solid-state electrochemical capacitor"; J. Mater. Chem. A, Vol. 2, pp. 1448-1457, 2014
- [187] Demetra S. Achilleos, T. Alan Hatton, "Surface design and engineering of hierarchical hybrid nanostructures for asymmetric supercapacitors with improved electrochemical performance", Journal of Colloid and Interface Science, Vol.447, pp. 282-301, 2015
- [188] Samia Mahouche-Chergui, <u>Sarra Gam-Derouich</u>, Claire Mangeney, and <u>Mohamed M. Chehimi</u>, "Aryl diazonium salts: a new class of coupling agents for bonding polymers, biomacromolecules and nanoparticles to surfaces", *Chem. Soc. Rev.*, Vol. 40, pp.4143-4166, 2011
- [189] Sümeyra Vural Kaymaz, Hediyeh, Malekzadsani Nobar, Hasan Sarıgül, Caner Soylukan, Lalehan Akyüz, Meral Yüce, "Nanomaterial surface modification toolkit: Principles, components, recipes, and applications", Advances in Colloid and Interface Science, Vol. 322, pp.103035, 2023
- [190] D. Lakshmi, M. Infanta Diana, P. Adlin Helen & P. Christopher Selvin, "Functionalized Nanomaterials, Classification, Properties, and Functionalization Techniques", <u>Functionalized Nanomaterials</u> <u>Based Supercapacitor</u> pp 65–92, 2024.
- [191] Zhenghui Li, Zhaopeng Li, Liuqing Li, Chengfei Li, Weihao Zhong, Haiyan Zhang, "Construction of Hierarchically One-Dimensional Core–Shell CNT@Microporous Carbon by Covalent Bond-Induced Surface-Confined Cross-Linking for High-Performance Supercapacitor", ACS Appl. Mater. Interfaces, Vol.9, Issue 18, pp.15557–15565, 2017
- [192] <u>Neha Singh</u>, <u>Ankur Malik</u>, <u>Shakshi Nohwar</u>, Rathin Jana, Prakash Chandra Mondal, "Covalent surface modification of nickel ferrite nanoparticles for electrochemical supercapacitor performance", New J. Chem, Vol. 47, pp. 5308-5315, 2023.
- [193] P. G. Campbell, M. D. Merrill, B. C. Wood, E. Montalvo, M. <u>A. Worsley</u>, T. F. Baumann, J Biener, 'Battery/supercapacitor hybrid *via* non-covalent functionalization of graphene macroassemblies", J. Mater. Chem. A, Vol. 2, pp. 17764-17770, 2014
- [194] Neeraj Kumar, Suprakas Sinha Ray, "Synthesis and Functionalization of Nanomaterials ",<u>Processing of Polymer-based</u> <u>Nanocomposites</u>, Vol.277, pp 15–55, 2018.
- [195] Mohammed Majdoub, Zakaria Anfar, Abdallah Amedlous, "Emerging Chemical Functionalization of g-C₃N₄: Covalent/Noncovalent Modifications and Applications", ACS Nano, Vol. 14, pp.10, 12390–12469, 2020.
- [196] Kwanchai Buaksuntear, Phakamat Limarun, Supitta Suethao, <u>Wirasak Smitthipong</u>, <u>Wirasak Smitthipong</u>, "Non-Covalent Interaction on the Self-Healing of Mechanical Properties in Supramolecular Polymers", *Int. J. Mol. Sci.* Vol. 23, Issue 13, pp.6902, 2022.

- [197] Salah Ud Din, Muhammad Sajid, Muhammad Imran, Javed Iqbal, Basit Ali Shah, Muhammad Azeem ullah Sufaid Shah, "One step facile synthesis, characterization and antimicrobial properties of Mg-doped CuO nanostructures", Mater. Res. Express, Vol. 6, pp 085022, 2019
- [198] Elena V. Shevchenko, Dmitri V. Talapin, Heimo Schnablegger, Andreas Kornowski, Örjan Festin, Peter Svedlindh, Markus Haase, Horst Weller, 'Study of Nucleation and Growth in the Organometallic Synthesis of Magnetic Alloy Nanocrystals: The Role of Nucleation Rate in Size Control of CoPt₃ Nanocrystals", J. Am. Chem. Soc. Vol. 125, Issue 30, pp. 9090–9101, 2003
- [199] Xinhang Guo, Siyang Lin, Yan Wang, Wangfeng Cai," Novel core-shell structure composite NiCo₂O₄-Vo@ZIF with amorphous ZIF shell and oxygen-vacancy-rich core for asymmetric supercapacitors", Electrochimica Acta, Vol. 475, 143636, 2024.
- [200] Ahmed Mohamed El-Toni, *^{ab} Mohamed A. Habila, ^e Joselito Puzon Labis, ^{ad} Zeid A. ALOthman, ^e Mansour Alhoshan, ^e Ahmed A. Elzatahry, Fan Zhang, "Design, synthesis and applications of core-shell, hollow core, and nanorattle multifunctional nanostructures", Nanoscale, Vol. 8, pp.2510-2531, 2016
- [201] Sattar Mohammadi Esfarjani, Ali Dadashi, Mohammad Azadi, "Topology optimization of additive-manufactured metamaterial structures: A review focused on multi-material types", Forces in Mechanics, Vol. 7,pp.100100, 2022
- [202] Lu Liu, Shuzhong Wang, Baoquan Zhang, Guanyu Jiang, Hui Liu, Jianqiao Yang, Jinglong Wang, Wei Liu,' Present status and prospects of nanostructured thermal barrier coatings and their performance improvement strategies: A review'', Journal of Manufacturing Processes, Vol. 97, pp.12-34,2023
- [203] R. Daniel, D. Holec, M. Bartosik, J. Keckes, C. Mitterer," Size effect of thermal expansion and thermal/intrinsic stresses in nanostructured thin films: Experiment and model", Acta Materialia, Vol. 59, Issue 17, pp. 6631-6645, 2011
- [204] Van-Nam Hoang, Phuong Tran, Van-Tuyen Vu, H. Nguyen-Xuan, "Design of lattice structures with direct multiscale topology optimization", Composite Structures, Vol. 252,112718, 2020.
- [205] Yi Gan, Cong Wang, Jingying Li, Junjie Zheng, Ziang Wu, Lin Lv, Pei Liang, Houzhao Wan, Jun ZhangHao Wang, "Stability Optimization Strategies of Cathode Materials for Aqueous Zinc Ion Batteries: A Mini Review", Front. Chem., Vol. 9.
- [206] Matthew, N. O'Brien, Matthew R. Jones, Chad <u>A. Mirkin chadnano, "The nature and implications of uniformity in</u> the hierarchical organization of nanomaterials", PNAS< Vol. 113, Issue 42, pp. 11717-11725, 2016
- [207] Zhengchen Wu, Han-Wen Cheng, Chen Jin, Bintong Yang, Chunyang Xu, Ke Pei, Huibin Zhang, Ziqi Yang, "Dimensional Design and Core–Shell Engineering of Nanomaterials for Electromagnetic Wave Absorption", Advanced Materials, Vol. 34, Issue11, pp. 2107538, 2022
- [208] Tian Gan, Zhikai Wang, Zhaoxia Shi, Dongyun Zheng, Junyong Sun, Yanming Liu, "Graphene oxide reinforced core-shell

structured Ag@Cu₂O with tunable hierarchical morphologies and their morphology–dependent electrocatalytic properties for biosensing applications", Biosensors and Bioelectronics Vol.112, pp.23-30, 2018

- [209] Qiannan Lin, Ziqi Sun, <u>Yuhai Dou</u>, <u>Jung Ho Kim, Shi Xue Dou</u>, "Two-step self-assembly of hierarchically-ordered nanostructures", J. Mater. Chem. A, Vol. 3, pp. 11688-11699, 2015.
- [210] Rongchen Shen^a, Chuanjia Jiang^b, Quanjun Xiang^c, Jun Xie^a, Xin Li, "Surface and interface engineering of hierarchical photocatalysts", Applied Surface Science, Vol. 471, pp. 43-87
- [211] Jian Sun, Mingrui Liang, Lu Yin, Geoffrey Rivers, Guangwei Hu, Qinmin Pan, Boxin Zhao, "Interfacial Compatibility of Core–Shell Cellulose Nanocrystals for Improving Dynamic Covalent Adaptable Networks' Fracture Resistance in Nanohybrid Vitrimer Composites", ACS Appl. Mater. Interfaces, Vol. 15, Issue 33, pp. 39786–39796, 2023
- [212] Yamei Li, <u>Shidong Ji</u>, Yanfeng Gao, Hongjie Luo, Ping Jin," Modification of Mott Phase Transition Characteristics in VO₂@TiO₂ Core/Shell Nanostructures by Misfit-Strained Heteroepitaxy", ACS Appl. Mater. Interfaces, Vol. 5, Issue 14, pp. 6603–6614, 2013
- [213] Ali Shafiee, Navid Rabiee, Sepideh Ahmadi, Marzieh Baneshi, Mehrdad Khatami, Siavash Iravani, Rajender S. Varma, "Core-Shell Nanophotocatalysts: Review of Materials and Applications", ACS Appl. Nano Mater. Vol. 5, Issue 1, pp. 55–86, 2022
- [214] Francesca Pietra' Luca De Trizio, <u>Anne W. Hoekstra' Nicolas</u> <u>Renaud' Mirko Prato' Ferdinand C. Grozema' Patrick J. Baesjou'</u> <u>Rolf Koole</u> Liberato Manna, Arjan J. Houtepen, "Tuning the Lattice Parameter of In_xZn_yP for Highly Luminescent Lattice-Matched Core/Shell Quantum Dots", ACS Nano, Vol. 10, Issue 4, pp. 4754–4762, 2016
- [215] JIATAO ZHANG, YUN TANG, KWAN LEE,
- AND <u>MIN OUYANG Authors Info & Affiliations</u>, "Nonepitaxial Growth of Hybrid Core-Shell Nanostructures with Large Lattice Mismatches", SCIENCE, Vol 327, Issue 5973, pp. 1634-1638, 2010
- [216] Jia Liu and Jiatao Zhang, "Nanointerface Chemistry: Lattice-Mismatch-Directed Synthesis and Application of Hybrid Nanocrystals", Chem. Rev. Vol. 120, Issue 4, pp. 2123–2170, 2020
- [217] panelJin Cao, Jinxu Liu, Xingwei Liu, Shukui Li, Xinying Xue, "Effect of the distribution state of transition phase on the mechanical properties and failure mechanisms of the W–Mo–Cu alloy by tuning elements content", Journal of Alloys and Compounds, Vol. 827, pp.154333, 2020
- [218] Zhirong Liao ^a, Andrea la Monaca, James Murray, Alistair Speidel, Dmitrii Ushmaev, Adam Clare, Dragos Axinte, Rachid M'Saoub, "Surface integrity in metal machining - Part I: Fundamentals of surface characteristics and formation mechanisms", International Journal of Machine Tools and Manufacture, Vol. 162,103687, 2021